# Transfer Adaptation Learning for Target Recognition in SAR Images: A Survey

Xinpeng Yang<sup>®</sup>, Lianmeng Jiao<sup>®</sup>, Member, IEEE, and Quan Pan<sup>®</sup>

Abstract—Synthetic aperture radar (SAR) target recognition is a fundamental task in SAR image interpretation, which has made tremendous progress with the advancement of artificial intelligence technology. However, SAR imaging is sensitive to the operating conditions of platforms, resulting in large distribution discrepancy for the data collected on different platforms. Moreover, SAR target images are difficult to annotate due to the blurry textures, resulting in insufficient labeled data to train a model. Therefore, subject to the data distribution discrepancy and insufficient labeled data, SAR target recognition becomes a highly challenging task. Transfer adaptive learning (TAL) is a learning paradigm aimed at completing target tasks by transferring knowledge from relevant source domains, which is a promising technique for solving the aforementioned problems in SAR target recognition. However, there is currently no comprehensive survey about the application of transfer adaptation learning in SAR target recognition. To this end, we comprehensively summarized the development of transfer adaptive learning in SAR target recognition, and provided systematic guidance for future research. In this article, we first summarized the electromagnetic features and visual features of SAR images used for target recognition, which can be potentially used for knowledge transfer. Then, we systematically reviewed the related literature according to the homogeneity of the transfer domains, the modality of the data in the source domain, and the category of the TAL methods. The available datasets that can be used to validate the TAL methods for SAR target recognition were also summarized for the researcher's convenience. We also conducted comparative experiments on these data to demonstrate the performance of TAL methods. Finally, we analyzed the main challenges of the current methods and pointed out several directions worth studying in the future.

*Index Terms*—Synthetic aperture radar (SAR), target recognition, transfer adaptation learning (TAL).

## I. INTRODUCTION

**I** N RECENT years, with the development of machine learning algorithms, the visual understanding of synthetic aperture radar (SAR) images has made tremendous progress [1], [2], [3]. Target recognition, which is a fundamental problem and holds a pivotal position in the visual understanding of SAR images, aims to recognize the fine-grained categories or type information of the target accurately [4], [5]. As shown in Fig. 1,

The authors are with the School of Automation, Northwestern Polytechnical University, Xi'an 710072, China (e-mail: xinpengyang@mail.nwpu.edu.cn; jiaolianmeng@nwpu.edu.cn; quanpan@nwpu.edu.cn).

Digital Object Identifier 10.1109/JSTARS.2024.3434448

Container Optical Target Recognition

Fig. 1. Difference between optical target recognition and SAR target recognition.

the SAR images always contain the speckle noise compared with the optical images due to electromagnetic wave imaging [6]. It is noteworthy that, given the reliance of SAR sensors on electromagnetic wave imaging, they remain unaffected by climatic variations and lighting conditions. Their extensive detection range and ability to penetrate vegetation further reinforce their advantages compared to optical target recognition [7], [8]. As a cutting-edge image interpretation method, SAR target recognition finds crucial applications in both military and civilian domains, ranging from disaster reporting and prevention to traffic management, urban planning, military intelligence gathering, etc., [9], [10]. SAR imaging is notably sensitive to the working conditions of platforms and the states of the target, including factors such as depression angle, polarization mode, wave band, and target azimuth [11], [12], [13]. Variations in these parameters significantly affect aspects of SAR images such as brightness, texture, scatter distribution, and noise levels. Therefore, in the task of SAR target recognition, if the training data and test data are collected from different platforms or under different working conditions, there will be significant differences in the data distributions between the training data and test data. This leads to the violation of the assumption made in traditional machine learning, which is that the samples in the training and test sample spaces are independent and identically distributed (I.I.D.) [14], [15]. For example, the images needed to be identified are collected after the working mode of radar has changed or from other platforms [16]. Traditional machine learning algorithms perform badly under these test conditions, as their generalization is poor.

One solution is to obtain and annotate data with the same distribution as the test data, and then retrain the model with supervised learning. However, it is time consuming and resource consuming. Whenever new test data are collected from other platforms or working conditions, the entire process needs to be

© 2024 The Authors. This work is licensed under a Creative Commons Attribution 4.0 License. For more information, see https://creativecommons.org/licenses/by/4.0/

Manuscript received 5 May 2024; revised 3 July 2024; accepted 22 July 2024. Date of publication 26 July 2024; date of current version 8 August 2024. This work was supported by the National Natural Science Foundation of China under Grant 62171386 and Grant 61801386 and in part by the Key Research and Development Program in Shaanxi Province of China under Grant 2022GY-081. (*Corresponding author: Lianmeng Jiao.*)

repeated, leading to unacceptable time and resource consumption. Meanwhile, annotating the SAR target images is extremely difficult for several reasons.

- SAR imaging relies on a distinct mechanism that involves active emission and reception of electromagnetic waves, which encompasses intricate physical processes such as electromagnetic scattering, interference, and diffraction. Consequently, SAR images exhibit different textures compared to optical natural images, deviating from human visual perception.
- SAR imaging is prone to various factors, including speckle noise, geometric deformation, and structural loss, leading to diverse visual representations of the same target type [7].
- 3) Moreover, SAR images of certain target models display a high degree of visual similarity, making them nearly indistinguishable to the human eye, even for experts with specialized knowledge in annotation.

A range of machine learning techniques aimed at improving the performance of SAR target recognition models but applied to different scenarios have also been employed, including metalearning techniques [17], [18], knowledge distillation [19], semisupervised learning [20], and unsupervised learning [21], etc. Specifically, metalearning techniques aim to uncover the inherent patterns and rules within the data itself, addressing the challenge of SAR target recognition with small sample sizes [17], [18]. On the other hand, knowledge distillation focuses on transferring the knowledge of larger, more complex models (teacher models) to smaller, more efficient models (student models), thereby enhancing the performance of SAR target recognition models with limited resources [19]. Semisupervised and unsupervised learning techniques aim to tackle SAR target recognition problems in various scenarios with varying degrees of labeled data availability. They rely on unlabeled or partially labeled data to learn meaningful representations and patterns, which can then be leveraged for recognition tasks [20], [21].

An effective way to solve the mentioned problem is to extract knowledge from other relevant data and apply it to SAR target recognition, which is called transfer learning or domain adaptation learning, and we use a general name transfer adaptation learning (TAL) for unifying them [22]. Transfer adaptation learning strives to acquire pertinent knowledge from source domain data that is associated with the target domain yet exhibits distinct distributions. The ultimate goal is to develop a model that excels at performing target domain tasks. A crucial aspect to note is that the target domain faces difficulties in directly crafting an effective model owing to the scarcity of annotated data, and so on. In contrast, obtaining annotated data in the source domain is comparatively easier, thereby bolstering the target domain task with pertinent knowledge. TAL imitates the learning ability of humans to draw analogies and to apply knowledge learned from other data. Due to these characteristics of TAL, it has widespread applications in natural language processing (NLP), computer vision (CV), speech signal recognition, fault diagnosis, and beyond [22]. In CV, TAL can be used for detection, recognition, and segmentation tasks under various cross-imaging conditions,



Fig. 2. Data types that can provide knowledge for SAR target recognition.

across different weather or lighting conditions, which are crucial for applications in autonomous driving, surveillance, and security [23]. TAL can also be applied to detection, recognition, and segmentation across different imaging modalities, such as pedestrian reidentification from optical to infrared, and medical image analysis from CT to MRI [24], [25], which are the key tasks in tracking, surveillance, and medical image analysis. TAL technology also has been widely utilized in the field of remote sensing [26], [27], [28], [29], including SAR target recognition [30], [31], [32], [33]. First, it leverages pretraining on extensive data closely related to the target domain, enabling the model to acquire and refine general features specific to SAR targets. Second, it minimizes the distribution discrepancy metrics between source and target domain SAR data by employing distribution alignment techniques. Third, it employs adversarial learning methods and other strategies to create shared feature spaces, thereby blurring the distinction between source and target domains. In the subsequent sections, we will delve deeper into the current TAL methodology tailored specifically for SAR target recognition.

To our knowledge, research on SAR target recognition based on TAL began in 2018 [34]. With the development of artificial intelligence technology, this research field is currently receiving increasing attention [35]. Up to now, there have been some surveys on the TAL [22], [36], [37], SAR target recognition [35], [38], [39], [40], or TAL in other fields [41]. As we can see the details of these surveys in Table I, they are not concerned on the SAR target recognition based on TAL. As far as we know, our paper is the first survey specialized for the SAR target recognition based on TAL. Over 70 excellent methods were systematically reviewed in our paper. We hope it can be a valuable tool and guidance to researchers interested in this field, assisting them in mastering the development direction.

For the SAR target recognition task, there are various types of data that can be taken as source domains to provide transferable knowledge, and each of them has its own characteristics, such as SAR images [30], simulated SAR images [31], optical images [32], and AIS [33], which can be seen in Fig. 2. Therefore, even the general methods have specific settings for different types of data, much less the methods that are designed for the special type of data. Hence, one can see that the literature categorization only based on the categories of methods is not suitable for SAR target recognition based on TAL. At the same time, the

Year	Author	Title	Content			
Survey	s on TAL					
2022	Zhang et al. [22]	Transfer Adaptation Learning: A Decade Survey	A systematic review of generic transfer learning technology.			
2022	Zhao et al. [36]	A Review of Single-Source Deep Unsupervised Visual Domain Adaptation	An overview of recent developments in deep unsupervised visual domain adaptation for natural optial images.			
2023	Bao et al. [37]	A Survey of Heterogeneous Transfer learning	A comprehensive review of generic transfer learning technology for heterogeneous domains.			
Survey	s on SAR target recogn	ition				
2016	El-Darymli et al. [38]	Automatic Target Recognition in Synthetic Aperture Radar Imagery: A State-of-the-Art Review	A comprehensive review of SAR target recognition from traditional methods to artificial intelligence methods.			
2018	Dong et al. [40]	A Review of the Autoencoder and Its Variants: A Comparative Perspective from Target Recognition in Synthetic-Aperture Radar Images	An overview of autoencoders applied to terrain classification, object detection, target recognition, and so on, in radar and hyperspectral images.			
2023	Li et al. [35]	A Comprehensive Survey on SAR ATR in Deep-Learning Era	Overview of the traditional and deep learning algorithms for SAR automatic target recognition.			
2023	Jiang et al. [39]	Radar Target Characterization and Deep Learning in Radar Automatic Target Recognition: A Review	A study on the radar target characteristic and deep learning methods used for radar target recognition.			
Survey	Surveys on TAL in other fields					
2024	Ma et al. [41]	Transfer learning in environmental remote sensing	A review on the transfer learning methods for environmental remote sensing research.			

acquisition principles of both SAR images and simulated SAR images are the same as those of the target domain, indicating that they are homogeneous data. Conversely, the acquisition principles of optical images and AIS data are significantly different from those of the target domain, indicating that they are heterogeneous data. Therefore, in our article, we first divide the literature into two top categories (homogeneous TAL and heterogeneous TAL) based on whether the modal of source data is the same as the target data, then divide the literature into four middle categories (from SAR to SAR, from simulated-SAR to SAR, from optical to SAR, from AIS to SAR) based on the types of data in the source domain, and finally divide them into bottom categories (pretraining, distribution-alignment, shared-featurespace) according to the categories of TAL methods. The detailed category information can be seen in Table II. Apart from this, the knowledge transferred and datasets used in these methods are also presented in columns 3 and 4 of this table. We believe that the multilevel literature categorization enables the readers to quickly and clearly find the relevant content they are interested in.

*Contributions:* Our article contributes to TAL in SAR target recognition by reviewing almost all relevant literature, analyzing the transferable features of SAR images, providing a detailed literature categorization, representing detailed information about available datasets, and offering suggestions and recommendations for future research. The key contributions are as follows.

- We investigated the effective electromagnetic and visual features for SAR target recognition and analyzed their potential applications for TAL in SAR target recognition, aiming to provide researchers with a comprehensive understanding of what kinds of knowledge can be transferred for SAR target recognition.
- 2) We have studied over 70 methods that apply TAL to SAR target recognition, and categorized these papers

systematically from top to bottom based on the homogeneity of the transfer domains, the modality of the data in the source domain, and the category of the TAL methods. Therefore, this article fills the gap in the current literature, which lacks a survey on the application of TAL in SAR target recognition.

- 3) We have investigated the datasets used in the reviewed literature, which could provide transferable knowledge for SAR target recognition. The detailed information and properties of these datasets are summarized. The experiments are also conducted on them to demonstrate the superiority of the TAL methods tailored specifically for SAR target recognition.
- 4) We have thoroughly analyzed the performance limitations of current methods and outlined potential directions for improvement. In addition, we have delved into unexplored scenarios in TAL for SAR target recognition, offering insights into promising future research directions.

*Organization:* The rest of this article is organized as follows. Section II introduces the preliminary knowledge of TAL and the categories of TAL methodology. Section III discusses transferable features of SAR images. Section IV reviews the homogeneous TAL methodologies, including transferring knowledge from SAR to SAR, from simulated-SAR to SAR. Section V reviews the heterogeneous TAL methodologies, including transferring knowledge from optical to SAR, from AIS to SAR, and from AIS and optical to SAR. Section VI gives the information of available data providing knowledge for SAR target recognition. Section VII provides comparative experiments to demonstrate the performance of TAL methods. Section VIII presents the discussions for existing methods and the recommendation for future work. Finally, Section IX draws the conclusions.

TABLE II SUMMARY OF REVIEWED LITERATURE

	References	Algorithms	Transferred Knowledge	Datasets
	From SAR to SAR	go	Interior Interior	
	2017 Huang et al. [30]	Pretraining	Model Weights	SAR Scene [30] $\rightarrow$ MSTAR [42]
	2017 Huang et al. [50]	Destesining	Model Weights	MSTAD [42] $\rightarrow$ OpenSADShin [44]
	2018 Zhang et al. [45]	Pretaining	Model Weights	MSTAR $[42] \rightarrow OpenSARShip [44]$
	2018 Shang et al. [45]	Pretraining	Model weights	MSTAR [42]
	2019 Zhang et al. [46]	Pretraining	Model Weights	SAR Scene [46] $\Rightarrow$ MSIAR [42]
	2019 Zhai et al. [47]	Pretraining	Model Weights	MSTAR [42]
	2020 Zhai et al. [48]	Pretraining	Model Weights	MSTAR [42]
	2022 Zhang et al. [49]	Pretraining	Model Weights	GPED [49] $\Leftrightarrow$ MTDD [50]
	2022 Wang et al. [51]	Pretraining	Model Weights	MSTAR [42]
	2022 Zhu et al. [52]	Pretraining	Model Weights	MSTAR [42]
	2022 Liu et al. [53]	Pretraining	Model Weights	Stylied-MSTAR [53] $\Rightarrow$ MSTAR [42]
	2022 Liu et al. [54]	Pretraining	Model Weights+ASC	MSTAR [42]
	2023 Pei et al. [55]	Pretraining	Model Weights	MSTAR [42] $\Rightarrow$ OpenSARShip [44]
	2020 Huang et al. [56]	Distribution-Alignment	Deep Feature	SAR Scene [56], MSTAR [42] $\Rightarrow$ OpenSARShip [44]
	2021 Zhao et al. [57]	Distribution-Alignment	Deep Feature	MSTAR [42]
	2022 Zhang et al. [58]	Distribution-Alignment	Deep Feature	MSTAR [42]
	2022 Chen et al. [59]	Distribution-Alignment	Deep Feature	MSTAR [42]
	2022 Zhang et al. [60]	Distribution-Alignment	Deep Feature+SR	MSTAR [42]. OpenSARShip [44]
د.	2020 Sun et al [61]	Shared-Feature-Space	Deep Feature	MSTAR [42]
(Y)	2022 Zhao et al [14]	Shared-Feature-Space	Deen Feature	SAR Shin [14]
E s	2022 Zhao et al. [11]	Shared-Feature-Space	Deep Feature	FUSAR [62] $\Leftrightarrow$ OpenSARShip [44]
no	2022 Zhao et al. [15]	Shared-Feature-Space	Deep Feature	$FUSAR [62] \Leftrightarrow OpenSARShip [44]$
ne	2022 Zhao et al. [10]	Shared Feature Space	Deep Feature	$MSTAP [42] \Leftrightarrow OpenSARShip [++]$
3ge	2022 Zhang et al. [05]	Shared Feature Space	Support Tensor	MSTAD [42]
ŭ	From Simulated-SAR to SA	P	Support Tensor	MSTAR [42]
Η	2017 Malmaren et al. [31]	Pretraining	Model Weights	SARSIM [65] $\rightarrow$ MSTAR [42]
	2018 Liu et al [66]	Pretraining	Model Weights	Airplanes Simulation [66] $\rightarrow$ SAR Airplanes [66]
	2019 Ma et al [67]	Pretraining	Model Weights	Vehicle GAN [67] $\rightarrow$ MSTAR [42]
	2019 Interest in [07]	Pretraining	Model Weights	SAMPLE [69] $\rightarrow$ MSTAR [42]
	2020 Liang et al [70]	Pretraining	Model Weights	Vehicle Simulation [70] $\Rightarrow$ MSTAR [42]
	2023 Zhang et al. [70]	Pretraining	Model Weights	SARSIM [65] SAMPLE [69] $\rightarrow$ MSTAR [42]
	2023 Kim et al [72]	Pretraining	Model Weights	SAMPLE [69] $\Rightarrow$ MSTAR [42]
ŀ	2023 Run et al [72]	Distribution-Alignment	Deen Feature	Vehicle Simulation [73] $\rightarrow$ MSTAR [42]
	2021 Tai et al. [74]	Distribution-Alignment	Deep Feature	Vehicle Simulation [74] $\Rightarrow$ MSTAR [42]
	2022 Tai et al [75]	Distribution-Alignment	Deep Feature	Vehicle Simulation [75] $\Rightarrow$ MSTAR [42]
	2022 He et al. [76]	Distribution-Alignment	Deep Feature	Vehicle Simulation [76] $\Rightarrow$ MSTAR [42]
	2023 Sun et al [77]	Distribution-Alignment	Deep Feature	SAMPLE [69] $\Rightarrow$ MSTAR [42]
	2023 Ly et al. [78]	Distribution-Alignment	Deep Feature	Vehicle Simulation [78] $\Rightarrow$ MSTAR [42]
	2023 Zhang et al. [79]	Distribution-Alignment	Deep Feature+ASC+SIFT	SAMPLE [69] $\Rightarrow$ MSTAR [42]
	2024 Han et al. [9]	Distribution-Alignment	Deep Feature	SAMPLE [69] $\Rightarrow$ MSTAR [42]
	2019 Zhang et al. [80]	Shared-Feature-Space	Deep Feature	Band1 Simulation [80] $\Rightarrow$ Band2 Simulation [80]
	2019 Wang et al. [81]	Shared-Feature-Space	Deep Feature	Vehicle Simulation [81] $\Rightarrow$ MSTAR [42]
	2021 Lv et al. [82]	Shared-Feature-Space	Deep Feature	Vehicle Simulation [82] $\Rightarrow$ MSTAR [42]
	2022 Du et al. [83]	Shared-Feature-Space	Deep Feature	Vehicle Simulation [83] $\Rightarrow$ MSTAR [42]
	2022 Chen et al. [12]	Shared-Feature-Space	Deep Feature	SAMPLE [69] $\Rightarrow$ MSTAR [42]
	2024 Shi et al. [84]	Shared-Feature-Space	Deep Feature+ASC	SAMPLE [69] $\Rightarrow$ MSTAR [42]
	From Optical to SAR			
	2018 Mufti et al. [34]	Pretraining	Model Weights	ImageNet [85] $\Rightarrow$ MSTAR [42]
	2018 Wang et al. [86]	Pretraining	Model Weights	ImageNet [85] $\Rightarrow$ SAR Ship [86]
	2019 Lu et al. [87]	Pretraining	Model Weights	ImageNet [85] $\Rightarrow$ SAR Ship [87]
	2019 Zhong et al. [88]	Pretraining	Model Weights	ImageNet [85] $\Rightarrow$ MSTAR [42]
	2020 Ying et al. [89]	Pretraining	Model Weights	ImageNet [85] $\Rightarrow$ MSTAR [42]
	2020 Taufique et al. [90]	Pretraining	Model Weights	Optical Ship [91] $\Rightarrow$ SAR ship [92]
	2021 Praneetha et al. [93]	Pretraining	Model Weights	ImageNet [85] $\Rightarrow$ MSTAR [42]
	2021 Relekar et al. [94]	Pretraining	Model Weights	ImageNet [85] $\Rightarrow$ OpenSARShip [44]
	2021 Huang et al. [95]	Pretraining	Model Weights	ImageNet [85] $\Rightarrow$ RESISC [96] $\Rightarrow$ MSTAR [42]
Т	2022 Tienin et al. [97]	Pretraining	Model Weights	ImageNet [85] $\Rightarrow$ Airbus [98]+SAR ship [99]
T	2022 Wang et al. [100]	Pretraining	Model Weights	ImageNet [85] $\Rightarrow$ MSTAR [42]
IS	2022 Lang et al. [101]	Pretraining	Model Weights	Optical Ship [102] $\Rightarrow$ SAR Ship [103], FUSAR [62]
100	2022 Tai et al. [104]	Pretraining	Model Weights	Optical Ship [91] $\Rightarrow$ SAR Ship [105]
gei	2022 Tai et al. [106]	Pretraining	Model Weights	Optical Vehicle [107] $\Rightarrow$ MSTAR [42]
ero	2023 Gao et al. [108]	Pretraining	Model Weights	$FGSCR [109] \Rightarrow FUSAR [62]$
lete	2019 Mohammad et al. [32]	Distribution-Alignment	Deep Feature	Optical Ship [91] $\Rightarrow$ SAR Ship [92], [105]
Ŧ	2022 Jahan et al. [110]	Distribution-Alignment	Deep Feature	Optical Vehicle [111] $\Rightarrow$ SAR Vehicle [111]
	2022 Zhao et al. [112]	Distribution-Alignment	Deep Feature	Optical Ship $[112] \Rightarrow$ FUSAR $[62]$
	2022 Song et al. [113]	Shared-Feature-Space	Deep Feature	$HRSC2016 [114] \Rightarrow HRSID [115]$
	From AIS to SAR	Destasiains	N 4 - J - 1 XX7- : - 1 ·	ALC [22] . CAD Chin [102]
	2018 Lang et al. [33]	Pretraining	Model Weights	AIS $[33] \Rightarrow 3AK 3 mp [103]$ AIS $[116] \rightarrow HD SAD [116] EUSAD [63]$
	2022 ran et al. [110] 2019 Xu et al. [117]	Distribution Alignment	NGFs	$\frac{\text{AIS [110]} \Rightarrow \text{figs} \text{AIS [110], FUSAK [02]}}{\text{AIS [33]} \Rightarrow \text{SAR Ship [103]}}$
	2017 Xu et al [117] 2021 Xu et al [118]	Distribution-Alignment	NGFs	AIS [33] $\Rightarrow$ SAR Shin [103]
	2022 Yang et al. [119]	Distribution-Alignment	Deep Feature	AIS $[33] \Rightarrow$ SAR Ship [103]
	From Optical and AIS to SA	AR		() · ·································
	2022 Lang et al. [102]	Shared-Feature-Space	Support Vector	AIS [33]+Optical Ship [102] $\Rightarrow$ FUSAR [62]

#### **II. PRELIMINARIES**

## A. Notations and Problem Definition

1) Notations: A summary of notations used in this article is given as follows to simplify understanding for readers.

$D_S$ :	Source Domain
$D_T$ :	Target Domain
$d_S$ :	Feature size of samples in source domain
$d_T$ :	Feature size of samples in target domain
$n_S$ :	Number of samples in source domain
$n_T$ :	Number of samples in target domain
$n_{TL}$ :	Number of labeled samples in target domain
$n_{TU}$ :	Number of unlabeled samples in target domain
$x_S \in \mathbb{R}^{d_S}$ :	Instance in source domain
$\mathcal{X}_S$ :	Feature space of instance in source domain
$y_S$ :	Label of instance in source domain
$\mathcal{Y}_S$ :	Label space of instance in source domain
$x_T \in \mathbb{R}^{d_T}$ :	Instance in target domain
$x_{TL} \in \mathbb{R}^{d_T}$ :	Labeled instance in target domain
$x_{TU} \in \mathbb{R}^{d_T}$ :	Unlabeled instance in target domain
$\mathcal{X}_T$ :	Feature space of instance in target domain
$y_T$ :	Label of labeled instance in target domain
$\mathcal{Y}_T$ :	Label space of instance in target domain

2) Problem Definition: According to the definition in the previous work [37], [120], [121], the domain consists of feature space  $\mathcal{X}$  and marginal probability distribution P(x), where  $x \in \mathcal{X}$  and follows P(x). The task  $\mathcal{T}$  based on the domain consists of the label space  $\mathcal{Y}$  and conditional probability distribution P(y|x), where  $y \in \mathcal{Y}$  and follows P(y|x). Therefore, the data in source domain are denoted as  $D_S = \{(x_S^i, y_S^i)\}_{i=1}^{n_S}$  where  $x_S \in \mathcal{X}_S$ , and  $y_S \in \mathcal{Y}_S$ . Different from the source domain, the data in target domain consists of two parts,  $D_T = D_{TL} \bigcup D_{TU}$ , which are respectively denoted as  $D_{TL} = \{(x_{TL}^i, y_{TL}^i)\}_{i=1}^{n_{TL}}$  and  $D_{TU} = \{x_{TU}^i\}_{i=1}^{n_{TU}}$ , where  $x_{TL}, x_{TU} \in \mathcal{X}_T$ , and  $y_T \in \mathcal{Y}_T$ . Under normal conditions,  $0 < n_{TL} \le n_{TU} \ll n_S$ .

Due to the discrepancy between source domain  $D_S$  and target domain  $D_T$ , which is caused by different work conditions, platforms, depression angles, polarization mode, wave band, azimuth, and so on, in SAR field, the marginal probability distribution and the conditional probability distribution of source and target domain are different,  $P_S(x) \neq P_T(x)$  and  $P_S(y|x) \neq P_T(y|x)$ . That is also the root reason why the methods designed for the source domain cannot generalize well on the target domain. TAL aims to solve this problem by transferring knowledge from the task  $\mathcal{T}_S$  on the source domain  $D_S$  to the task  $T_T$  on the target domain  $D_T$ . Specifically, given the data of source domain  $D_S = \{(x_S^i, y_S^i)\}_{i=1}^{n_S}$ , the task  $\mathcal{T}_S$ , and the data of target domain  $D_T = \{(x_{TL}^i, y_{TL}^i)\}_{i=1}^{n_{TL}} \bigcup \{x_{TU}^i\}_{i=1}^{n_{TU}}$ , that is to say, given the marginal probability distributions  $P_S(x)$  and  $P_T(x)$ , and the conditional probability distribution  $P_S(y|x)$  and  $P_{TL}(y|x)$ , the goal of TAL is learning the objective conditional probability distribution  $P_{TU}(y|x)$ . There are some special circumstances: when  $n_{TL} > 0$ , it is always tagged with semisupervised learning; when  $n_{TL} = 0$ , it is tagged with unsupervised learning.



Fig. 3. Paradigm of pretraining.



Fig. 4. Distribution discrepancy between domains.

#### B. Categories of TAL Methodology

Regardless of the modality of data used as the source domain for knowledge transfer, the literature reviewed in this article can be divided into three categories based on the techniques used, namely: Pretraining, distribution alignment, and shared feature space. In this section, we introduce the general forms of these methods, and the methods designed specifically based on the data characteristics will be reviewed in detail in the following sections.

1) Pretraining: Datasets with large volumes contain rich and diverse data, which can provide the general knowledge for target recognition or other visualization task. Pretraining the model on these datasets enables the model to extract general features, and then the model can be customized for the downstream task by fine-tuning operation, which refer to the task of training a model in specific application scenarios, such as target recognition, detection, or segmentation [34]. The paradigm of pretraining and fine-tuning can be seen in Fig. 3. This method is simple to implement but relies on the dataset with a large amount of labeled data.

2) Distribution Alignment: Due to the difference between imaging environments, platforms, lighting conditions, and so on, the source training data and the target test data show different probability distributions, which can be seen in Fig. 4. The manifestation of distribution differences in image data is that the targets in images have different appearances [56]. Distribution alignment methods decrease the distribution differences by minimizing the distribution discrepancy metrics of the features from source and target, such as MMD [56], LMMD [122], KLD [110], and the paradigm of the distribution alignment can



Fig. 5. Paradigm of distribution alignment.



Fig. 6. Paradigm of adversarial learning.

be seen in Fig. 5. This method is effective but relies on predefined distribution discrepancy metrics.

3) Shared Feature Space: Different from the pretraining methods relying on sufficient data and the distribution alignment methods relying on the distribution discrepancy metrics, building a feature space where the features come from different domains cannot be discriminated against is also an effective paradigm, which is called the shared feature space method. The diagram of the commonly used method to build shared feature space, adversarial learning, which endeavors to confuse the feature representations of data from distinct domains through a competitive game between feature extractors and domain discriminators, can be seen in Fig. 6. The shared feature space method confuses the features from source and target domains so that the classifier can perform well regardless of the domains.

#### **III. TRANSFERABLE FEATURES OF SAR IMAGES**

Relatively speaking, the deep visual features of SAR images have been extensively studied in TAL for SAR target recognition, as outlined in the reviewed methods. However, the special property features of SAR images, such as electromagnetic scattering features and traditional features, have not been fully utilized in the TAL for SAR target recognition [54], [79]. During the review process, it was found that some work had fused these features with deep features in SAR recognition tasks for TAL to improve performance [60], [79], [84], [117]. This indicates that this approach is able to improve the effectiveness of TAL by introducing additional information. Consequently, they are highly valuable considerations for future endeavors and can be effectively utilized in TAL for SAR target recognition, thereby improving both model performance and generalization capabilities. Therefore, they are the potential sources of knowledge transfer in SAR target recognition. In this section, we discuss the traditional features that may be useful for knowledge transfer in SAR target recognition, such as the electromagnetic scattering features extracted based on the attribute scattering center (ASC) model [123], the visual features such as SAR-SIFT [124], histogram of oriented gradient (HOG) [125], wavelet transformation (WT) [126], sparse representation (SR) [127], Naive geometric features (NGFs) [128]. Electromagnetic features stem directly from the principles of SAR imaging. They predominantly capture intrinsic physical attributes related to the target's surface material composition, such as scattering patterns, reflectivity, and polarization characteristics. These features are inherently linked to the electromagnetic interactions between the radar wave and the target, providing insights into the target's material composition and geometric shape. In contrast, visual features are extracted directly from the SAR images. They encompass attributes related to the image's texture, pixel statistics, pixel variation trends, target size, and other transformations of the image data. Visual features capture the visual appearance and spatial relationships of objects within the SAR image. Both categories of features possess their unique value in SAR target recognition, offering complementary information that can be integrated to optimize the performance of recognition models. In contrast, the deep features suffered the question about network structure modifications, optimization tricks, loss function designing, and so on, bringing challenges for further performance improvements. Therefore, the traditional features should not be abandoned [129].

## A. Attribute Scattering Center

The ASC model was first proposed by Gerry et al. [123], which extended the physical optical theory and geometrical theory of diffraction [130]. It denotes that, under high-frequency conditions, the electromagnetic scattering is approximate to the superposition of a series of individual electromagnetic scattering centers [5]. ASC model can be described by a set of functions of operation frequency and aspect angle of radar, which is formulated as

$$E(f,\phi;\Theta) = \sum_{i=1}^{n} E_i(f,\phi;\Theta_i)$$
(1)

where  $E(f, \phi; \Theta)$  denotes the total electromagnetic scattering,  $f, \phi$  denotes the operation frequency and aspect angle of radar,  $\Theta$  denotes the physical parameters, and  $E_i(f, \phi; \Theta_i)$  denotes the *i*th electromagnetic scattering center. In details, the *i*th electromagnetic scattering center can be described as follows:

$$E_{i}(f,\phi;\Theta_{i}) = A_{i} \cdot \left(j\frac{f}{f_{c}}\right)^{\alpha_{i}}$$
$$\cdot \exp\left(\frac{-j4\pi f}{c}\left(x_{i}\cos\phi + y_{i}\sin\phi\right)\right)$$
$$\cdot \sin c\left(\frac{2\pi f}{c}L_{i}\sin\left(\phi - \bar{\phi}_{i}\right)\right)$$
$$\cdot \exp\left(-2\pi f\gamma_{i}\sin\phi\right) \tag{2}$$

TABLE III DETAILED INFORMATION OF GEOMETRIC SCATTERING TYPES

	Geometric Scattering Type	$\alpha$	$\gamma$	L	$\bar{\phi}$
Trihedral Top Hat Sphere		$\begin{array}{c}1\\0.5\\0\end{array}$	> 0	0	0
	Corner Diffraction	-1			
	Dihedral Cylinder	1 0.5			$\neq 0$
Distributed	Edge Broadside Edge Diffraction	0 -0.5	0	> 0	

where c denotes the speed of electromagnetic wave,  $f_c$  denotes the radar central frequency,  $\Theta_i = [A_i, x_i, y_i, \alpha_i, L_i, \overline{\phi}_i, \gamma_i]$  is the physical parameters of the *i*th electromagnetic scattering center, in which  $A_i$  represents the amplitude of it,  $(x_i, y_i)$  represents the geometrical location,  $\alpha_i$  denotes the frequency dependence,  $L_i$ denotes the length,  $\phi_i$  denotes the azimuth, and  $\gamma_i$  denotes the azimuth dependence. The ASC model contains intrinsic property information of the radar target, including electromagnetic diffraction resulting from sharp or protruding parts of the target. This diffraction reflects the physical structure of the target and facilitates target recognition, and the information of the most common scattering structures can be seen in Table III. Based on the ASC model and optimization algorithms, the parameters representing the property information of the radar target can be estimated. Then, electromagnetic features can be obtained through feature extraction methods based on the estimated parameters, and widely utilized in SAR target recognition [5], [131], [132], [133]. Some researchers have also applied the information directly related to the radar target itself, obtained based on the ASC model, to the TAL task of SAR target recognition. They use feature extraction methods to obtain electromagnetic scattering features and fuse them with the visual features of SAR images to enhance the generalization of the model [54], [79], or integrate this information as prior knowledge into the training process of the model to improve its recognition performance [84]. It can be seen that, by integrating SAR target electromagnetic scattering knowledge with the recognition model, the generalization of the model can be effectively improved. Therefore, how to transfer electromagnetic scattering knowledge based on the ASC model is a question worth exploring.

#### B. SAR-SIFT

SAR-SIFT is one of the most common keypoints detection and scale-invariant feature extraction algorithms that are used for the SAR image interpretation [124]. SAR SIFT can avoid the influence of speckle noise to a considerable extent, extracting scale-invariant, rotation-invariant, and translation-invariant features. It obtains keypoints and descriptors that encapsulate gradient information from surrounding pixels, with the ultimate goal of reflecting edge information through gradient computation. SAR-SIFT is also used to build features for SAR target detection [134] and recognition [135], [136]. Therefore, the SAR-SIFT features are promising for TAL in SAR target recognition.

### C. SAR-HOG

The HOG feature was first proposed for person detection, which is not sensitive to local geometric transformations, such as translations or rotations [125]. Considering that SAR images are sensitive to the depression angle and the azimuths of targets, Song et al. [137] proposed SAR-HOG, which pays more attention to the stable SAR image pixels that are reflected by strong backscatter returns from structures exhibiting aspect insensitivity. After that, the HOG feature is widely used for SAR image interpretation [138], [139], including SAR target recognition [129], [140], [141]. Therefore HOG features have the potential to be fused with deep features for TAL to improve the SAR target recognition performance.

## D. Wavelet Transformation

WT is developed based on the Fourier transformation. Fourier transformation is invariant for translation, but variant for deformation of the high frequency parts [126]. Generally speaking, image features ought to exhibit invariance to geometric transformations, encompassing translation, rotation, scale transformation, and minor deformations, while also possessing robustness against perturbations. The WT features possess these properties but are sensitive to translation. Therefore, some researchers have improved it and applied it to SAR target recognition [142] or combined deep feature [126], [143], [144]. WT features have shown good performance in SAR target recognition tasks and can be further fused with deep features. It is also worth considerable for TAL in this field.

## E. Sparse Representation

SR is widely used for face recognition [145], image denoising [146], and SAR target recognition [103], [147], [148]. The basic idea of the SR classification is to represent the test sample with a linear combination of training samples. The category of the test sample can be determined by the sparse reconstruction error [148]. The SR has good anti-noise performance, and is widely used for SAR target recognition [149], [150], [151] or combined deep features [60]. SR is complementary with deep features in SAR target recognition to some extent, and it should be considered in TAL for SAR target recognition [60].

## F. Naive Geometric Features

NGFs are used for ship classification in SAR images [128], [152], which represent the geometric properties of the ships, such as length, width, perimeter, and area ratio. The detailed information about NGFs can be seen in Table IV. For SAR target images, these features can be obtained by calculating the minimum bounding boxes surrounding the target [33], [116]. The experiments [128], [152] in SAR images show that the NGFs are effective features for ship fine-grained recognition. It can be seen that these features are common in some different types of data. Therefore, they are also worth considering for TAL in SAR target recognition [128], [152].

In this section, we discussed the traditional features used in the reviewed literature and presented several potential features that

TABLE IV NAIVE GEOMETRIC FEATURES

Features Description		Definition	
$f_1$	length		
$f_2$	width	W	
$\bar{f_3}$	naive perimeter	$2 \times (L+W)$	
$\bar{f_4}$	naive area	$L \times W$	
$f_5$	aspect ratio (a)	L/W	
$f_6$	aspect ratio (b)	W/L	
$f_7$	shape complex	$(L+W)^2/(L\times W)$	
$f_8$	/	$W^2/(L^2+W^2)$	
$f_{9}$	/	(L - W)/(L + W)	
$f_{10}$	/	L/(L+W)	
$f_{11}^{10}$	/	W/(L+W)	

could be applied in TAL for SAR target recognition. However, the manners of utilizing these features remain worth exploring in this field. The features that can be used in TAL for SAR target recognition are not limited to these, and we should explore more potential features in the future and consider how to utilize them effectively.

#### IV. HOMOGENEOUS TRANSFER ADAPTATION LEARNING

SAR and simulated-SAR images are two kinds of homogeneous data that can provide knowledge for SAR target recognition. Without a doubt, the source domain SAR images and target domain SAR images are absolutely homogeneous, with only existing discrepancies between operation conditions or platforms. Differently, simulated-SAR images are obtained by a simulation system based on electromagnetic theory and target modeling, which is similar to the process of SAR imaging [31]. Therefore, we refer to the methods using homogeneous data in both the source and target domains as homogeneous TAL. The simulated-SAR images can be more easily obtained than SAR images, but SAR images are more in line with real scenarios. Therefore, how to transfer knowledge from these data to SAR target recognition has become a hot topic.

#### A. From SAR to SAR

Transferring knowledge from SAR images to SAR images can be the most direct method in homogeneous TAL for SAR target recognition. There are two kinds of SAR images, scene images and target images. The quantity of the SAR scene images is always larger than the SAR target images. However, annotating scenes or target images is time-consuming and source-consuming. How to transfer the imaging modal knowledge (information of SAR images, regardless of contents) from SAR scene images is a question worth of exploring, and many kinds of methods are proposed to solve this [14], [15], [16], [30], [43], [45], [46], [47], [48], [49], [51], [52], [53], [54], [55], [56], [57], [58], [59], [61], [63], [64], [153].

1) Pretraining Method: Pretraining and fine-tuning is a commonly used learning paradigm in this field. Based on the ability of the pretrained model to extract common features, the model can be easily adapted to downstream tasks, such as SAR target recognition [30], [43], [45], [46], [47], [48], [49], [51], [52], [53], [54], [55], [153].

The performance of directly utilizing the pretrained model is often unsatisfactory. Therefore, many technologies are combined to improve the performance of the SAR target recognition model. Considering that there is no well-annotated SAR target data can be used, Huang et al. [30] proposed to transfer knowledge from unlabeled SAR scene data with large volumes. A convolutional network was pretrained on the image reconstruction task to extract general features, and then fine-tuned on the SAR target recognition task. Zhang et al. [153] pretrained a generative adversarial network (GAN) on unlabeled SAR scene images as the initial model for the SAR target recognition task. Inspired by this, Zhang et al. [46] pretrained a combination of GAN and reconstruction network on the unlabeled SAR images to improve performance. Compared with pretraining on the unlabeled SAR scene images, pretraining on the SAR target images can be more direct for the recognition task. Zhang et al. [43] pretrained the network on the SAR vehicle images and then fine-tuned it on the SAR ship images, which transferred knowledge across tasks. Zhang et al. [49] considered that the number of ships in the SAR ship detection dataset is much larger than the SAR ship recognition dataset. They pretrained the network on the SAR ship detection dataset to extract the general features of ships and then fine-tuned it on the ship recognition dataset to achieve ship target recognition.

To improve the performance of pretraining methods, some researchers adapted data augmentation to enrich the diversity of data. Zhai et al. [47] employed geometric image transformation to generate new SAR vehicle images to enlarge the volume of the pretraining dataset. Liu et al. [53] adapted style transfer technology to extend MSTAR [42], then pretrained and fine-tuned the network for ship recognition. Pei et al. [55] combined the geometric transformation augmentation and contrastive learning to enable the model to extract discriminative features and then fine-tuned it for ship recognition.

Some well-designed modules are plugged into the network to improve the performance. Shang et al. [45] added an information recorder module and improved the generalization of the network with a hinge loss function. Zhai et al. [48] designed a multilevel feature fusion attention module to improve the discriminative of features. Wang et al. [51] pretrained a Siamese network to iteratively predict pseudolabel for unlabeled target images, and fine-tuned the network based on them.

To utilize the SAR imaging modality characteristics, Liu et al. [54] proposed a complex convolutional network with electromagnetic properties transfer. It calculated new convolutional kernels based on modified ASC model, so that obtained better discriminative features through pretraining.

Varieties of pretraining methods have been proven to be effective in the homogeneous TAL for SAR target recognition, but there are some questions worth considering:

- 1) The volume of the SAR dataset used for pretraining is too small compared with the common dataset in computer vision, for example, ImageNet [85].
- 2) These methods ignored the essential discrepancy between probability distributions of the source domain and target domain.

2) Distribution Alignment Method: It is assumed that the training and test data come from the same distribution in

traditional machine learning. However, the data distribution discrepancy caused by imaging conditions of SAR platforms has broken this assumption, resulting in poor generalization of the model trained in the source domain. Therefore, aligning distributions is a kind of effective method to improve the generalization of the network for SAR target recognition [56], [57], [58], [59].

Huang et al. [56] first and comprehensively explored the difference between networks, source tasks, network layers, and TAL methods for SAR target recognition. The multikernel maximum mean discrepancy (MK-MMD) was adopted to measure the discrepancy between the marginal distributions of source and target data, and transitive TAL was proposed to deliver knowledge across multisource (optical image  $\rightarrow$  SAR scene image  $\rightarrow$  SAR target image). The MK-MMD and whole loss function can be formulated as

$$\mathrm{MMD}(x_S, x_T) = [E_S \left[\phi(x_S)\right] - E_T \left[\phi(x_T)\right]]_{\mathcal{H}}$$
(3a)

$$\mathcal{L} = \mathcal{L}_C(x_{TL}, y_{TL}) + \lambda \sum_{l=k+1}^{L} \alpha_l \text{MMD}_l(x_S, x_T)$$
(3b)

where  $\phi$  denotes one of the functions in the unit of reproducing kernel Hilbert space (RKHS)  $\mathcal{H}, E[\cdot]$  denotes the mathematical expectation,  $\lambda$  denotes a tradeoff parameter, and  $\alpha_l$  denotes the weights of MMD loss for layer l. Their experiments proved that the images containing the targets of similar categories are more suitable for transferring knowledge, the shallow layers are easier to transfer, and the distribution alignment method is more effective than pretraining methods. Considering that directly performing distribution alignment will cause the domain special knowledge to be lost. Zhang et al. [58] designed a shared subnetwork and special subnetwork to deal with the domain common and private properties, and then performed MK-MMD to align the distributions. The unlabeled target domain data contain important information too, and the most direct method to make use of it is the pseudolabel strategy (training model based on the unlabeled data with predicted label). Chen et al. [59] performed weak image augmentation for pseudolabel predicting and strong image augmentation for performance improvement. They also aligned the distributions with MK-MMD, and decreased the negative impact of the incorrect pseudolabel by top-k loss [154]. However, the discrepancy between different categories is not explored in these method designs. Zhao et al. [57] predicted the pseudolabel and calculated class confusion loss [155] to align the conditional distributions for each category, which is formulated as

$$\mathcal{L}_{CC} = \frac{1}{c} \sum_{i=1}^{c} \sum_{j \neq i}^{c} \left| \frac{C_{ij}}{\sum\limits_{k=1}^{c} C_{ik}} \right|$$
(4)

where c denotes the number of class,  $C_{ij}$  denotes the weighted class correlation between class i and class j. In addition, Zhang et al. [60] conducted a preliminary exploration of online domain adaptation. They achieved good performance by aligning the distribution of target data with maximum a posteriori estimation, using the combination of deep features with SR features.

Although distribution alignment methods have achieved good performance, there are still some details that lack exploration.

- The metrics used to measure discrepancy between probability distributions often overlook the conditional distribution and covariance. Incorporating marginal and conditional probability distributions, as well as covariance, could lead to a more accurate alignment.
- The current distribution alignment methods fail to leverage properties specific to SAR targets, such as electromagnetic scattering characteristics. By utilizing these properties, the alignment process and performance could be potentially enhanced.

*3)* Shared Feature Space Method: The shared feature space methods are mainly concentrated on building a feature space, where the features come from different domains that cannot be discriminated against or where the target characteristics have a general representation regardless of domains. Many interesting methods have emerged to achieve this [14], [15], [16], [61], [63], [64].

The adversarial learning strategy is always used to build the shared feature space. Zhao et al. [14] confused the features of data from different platforms by adversarial learning to decrease the domain gap. Considering that learning from the hard samples is key to improving performance, Zhao et al. [16] proposed a dynamic hard sample selection method to increase the importance of hard samples in the adversarial learning process. To utilize the information from unlabeled target data, Zhao et al. [15] combined the pseudolabel strategy and adversarial learning, and they corrected the pseudolabel with a class confusion matrix. The whole loss function can be formulated as

$$\mathcal{L}_{adv} = E_S \left[ \log(\mathcal{F}_D^S \left( x_S \right)) \right] + E_T \left[ \log(1 - \mathcal{F}_D^T \left( x_T \right)) \right]$$
(5a)  
$$\mathcal{L} = \mathcal{L}_C(x_S, y_S) + \mathcal{L}_C(x_T, y_{TP}) + \lambda \mathcal{L}_{adv}$$
(5b)

where  $\mathcal{F}_D^T$  and  $\mathcal{F}_D^S$  represent the domain discriminator and  $y_{TP}$  denotes the pseudolabel for target samples. In addition, Zhang et al. [63] utilized adversarial learning to confuse the distributions of support set and query set in metalearning task, therefore improved the network's adaptability across different SAR target recognition tasks.

Different from building a shared feature space with adversarial learning, Gao et al. [64] proposed a method based on the support tensor machine. This method mapped SAR images into a shared feature space using a shared core tensor, thus confusing the two domains and decreasing the domain gap. However, considering that the properties of the target were not taken into account when building the shared feature space, Sun et al. [61] perceived that angular rotation is a common domain difference. They proposed an attribute-guided transfer learning method, which constructed a shared feature space where the original features could be transformed into features of any other azimuth, thereby decreasing the domain gap, which can be formulated as

$$\mathcal{L}_{r} = \|\mathcal{I}_{1} - \delta_{1}(x_{1})\| + \|\mathcal{I}_{2} - \delta_{1}(M_{g}x_{1})\|_{2}^{2}$$
(6a)

$$M_g x_1 = x_1 + \gamma \mathcal{R}\left(x_1\right) \tag{6b}$$

where  $\mathcal{I}$  denotes the original image,  $\delta$  denotes the reconstruct block,  $M_g$  denotes the feature transformation, and  $\gamma$  denotes the normalized angle difference.

The shared feature space methods have achieved good performance, but there are still some directions that need to be explored: 1) How to increase or maintain discrimination between different categories is not considered in these methods. 2) How to build a shared feature space according to the properties of SAR images and targets needs to be considered further.

#### B. From Simulated-SAR to SAR

Despite the fact that many effective methods have been proposed for TAL from SAR to SAR, the insufficient data problem is still significant in this field. To further improve the data-driven methods for SAR target recognition, transferring knowledge from the simulated SAR data is an alternative idea. With the auto-CAD models and electromagnetic scattering mechanism, a simulated SAR target image dataset with a large volume can be obtained using software [65], [156], [157]. However, as the simulated data are essentially produced from the ideal model, there are differences in detail from electromagnetic scattering in real radar systems. How to transfer knowledge from the simulated data is also an issue worth of research [9], [12], [31], [66], [67], [68], [70], [71], [72], [73], [74], [75], [76], [77], [78], [79], [80], [81], [82], [83], [84].

1) Pretraining Method: The category of the target is exactly known in the simulation process, and it is easy to obtain a large amount of annotated simulated SAR target data. Pretraining methods are the most direct to utilize these simulation data [31], [66], [67], [68], [70], [71], [72].

Some methods directly pretrained the model on the simulated data to obtain the optimal initialization weights [31], [67], [68], [70]. Ma et al. [67] produced the simulated data by LSGANs instead of the simulation software, and they specialized the network for recognition tasks by fine-tuning it on the real SAR data. Inkawhich et al. [68] also discussed the influence of data augmentation, model construction, loss function choices, and ensembling techniques in pretraining model training.

Considered that there is still a domain gap between simulated data and real data due to the difference between simulation and real radar environments. Some methods processed the simulated data further [66], [71], [72]. They generalized intermediate data with more similarity to the real SAR data from simulated SAR data using CycleGAN or conditional-GAN, thus decreasing the discrepancy between simulated data and real data. They all followed the learning paradigms of pretraining on the simulated SAR data and fine-tuning on the real SAR data.

Whether one chooses to directly pretrain or to further process the simulated SAR data, there are some questions that need to be further considered: 1) The discrepancy between probability distributions needs to be taken into account. 2) The electromagnetic scattering knowledge, which is crucial for simulated SAR data, requires discussion.

2) Distribution Alignment Method: Due to the fact that simulation data are obtained in an ideal environment, there



Fig. 7. Paradigm of metric learning across domains.

is an unavoidable distribution discrepancy between simulated and real SAR data. Some distribution alignment methods are proposed to solve this problem [9], [73], [74], [75], [76], [77], [78], [79].

In these methods, MMD is still the most popular metric to measure the discrepancy between data distributions [73], [76], [77]. Especially, in addition to decreasing the domain gap by minimizing the MMD, Sun et al. [73] ensembled a multiscale subclassifier and obtained the final output using a voting strategy. Furthermore, Sun et al. [77] performed pseudolabel generation and denoising, and fine-tuning of the classified network. Han et al. [9] presented evidential learning to estimate the confidence degree of the model, dynamic weighting to avoid the impact of inferior knowledge, and implemented knowledge distillation from simulation to real SAR target recognition by minimizing KLD.

To further align the conditional distributions for each category, Lv et al. [78] proposed to minimize the LMMD [122] combined image reconstruction task. The trained network could extract discriminative features from both simulated and real SAR images, thus improving the performance further. The LMMD and total loss can be formulated as

$$LMMD(x_S, x_T) = E_c \|E_{S^c}[\phi(x_S)] - E_{T^c}[\phi(x_T)]\|$$
(7a)  
$$\mathcal{L} = \mathcal{L}_C (x_S, y_S) + \mathcal{L}_C (x_{TL}, y_{TL}) + LMMD(x_S, x_T).$$
(7b)

The intra- and interclass metric learning is a method to implicitly align the conditional probability distributions, which minimizes the distance between samples in the same class and maximizes the distance between samples in different classes regardless of the domains [74], [75], which can be seen in Fig. 7. Tai et al. utilized the d-SNE loss [158] to achieve cross-domain intra- and interclass metric learning. At the same time, Tai et al. [74], [75] considered that the samples of partial azimuths cannot comprehensively provide discriminative information of targets. They proposed to extract angle invariant properties by learning feature translation between samples with different azimuths [74] and conducting synthetic samples with different azimuths by condition GAN [75]. The d-SNE and whole loss can be formulated as

$$\mathcal{L}_{d-SNE} = \frac{1}{n_T} \sum_{i=1}^{n_T} \log \left( \frac{\sum\limits_{y_S^k \neq y_T^i} \exp\left(-d\left(x_s, x_T^i\right)\right)}{\sum\limits_{y_S^k = y_T^i} \exp\left(-d\left(x_s, x_T^i\right)\right)} \right)$$
(8a)  
$$\mathcal{L} = \mathcal{L}_{d-SNE} + \lambda \mathcal{L}_C(x_S, y_S) + \beta \mathcal{L}_C(x_{TL}, y_{TL}).$$
(8b)

Considering that the electromagnetic scattering features of SAR images are beneficial for SAR target recognition. Zhang et al. [79] proposed to obtain scattering points with the ASC model and SAR-SIFT algorithm, and extract scattering topological electromagnetic scattering features with two graph neural networks. More important, CORAL [159] was utilized to measure the distribution discrepancy with the second-order statistics, which is formulated as

$$\mathcal{L}_{\text{CORAL}}(x_S, x_T) = \frac{1}{4d_S^2} \|C_S - C_T\|_F^2$$
(9a)

$$C_{S} = \frac{1}{n_{S} - 1} \left( x_{S}^{T} x_{S} \right), C_{T} = \frac{1}{n_{T} - 1} \left( x_{T}^{T} x_{T} \right).$$
(9b)

The distribution alignment methods minimize the distribution discrepancy metrics to decrease the domain gap, with metrics playing a crucial role. 1) The predefined metrics may not be suitable for the scene of TAL from simulated SAR data to real SAR data. 2) The special properties of SAR images or radar targets, such as scattering features, are not explored enough.

*3)* Shared Feature Space Method: The core of shared feature space methods for TAL from simulated SAR data to real SAR data lies in creating a feature space, where simulation and real SAR images become indistinguishable based on their features. In this space, targets with similar attributes are represented by similar feature vectors [12], [80], [81], [82], [83], [84].

The idea of adversarial learning remains a classic one to build shared feature space for this scene [12], [80], [81], [82], [83]. Zhang et al. [80] proposed to decrease the gap between simulation images of different bands using adversarial learning to achieve SAR target recognition across frequency bands. Wang et al. [81] utilized the simulated SAR data to make up for the insufficient data problem in metalearning for SAR target recognition, and they confused simulated data and real data utilizing adversarial learning. Du et al. [83] successfully transferred knowledge from simulated to real SAR data using adversarial learning, while also incorporating image reconstruction to preserve crucial information and enhance performance.

To further decrease the domain gap between simulated SAR data and real SAR data, some methods take a step by distribution alignment or GAN [12], [82], [84]. Lv et al. [82] combined minimizing the MK-MMD and adversarial learning losses to transfer knowledge from simulated SAR vehicle image to real SAR vehicle image. Shi et al. [84] proposed a pseudolabel refinement strategy based on the ASC model. They also employed contrastive learning (learning the representation by maximizing the similarity between similar samples and minimizing the similarity between dissimilar samples) to compact intraclass prototypes and separate interclass prototypes, further narrowing the domain gap. Chen et al. [12] used CycleGAN to generate intermediate domain data bridging simulated and real data. Subsequently, they performed adversarial learning between the intermediate and real SAR domains, significantly reducing the complexity of the learning process and greatly enhancing performance.

The shared feature space methods aim to confuse the domains and thereby reduce the domain discrepancy. However, there are some questions needed to consider further: 1) How can we effectively perform adversarial learning while simultaneously maintaining discriminative information. 2) The common electromagnetic scattering properties or target characteristics shared between simulation and real data have not received sufficient attention.

## V. HETEROGENEOUS TRANSFER ADAPTATION LEARNING

Currently, there are two kinds of heterogeneous data that can provide transferable knowledge for SAR target recognition: optical images and AIS data. Optical imaging relies on receiving reflected light by optical sensors, which is passive imaging compared to SAR. Obviously, the optical images typically contain clearer textures and less noise, but are susceptible to lighting conditions [32]. AIS, designed specifically for vessels, is a real-time network that combines transmitters and receivers to transport dynamic, static, and voyage-related text information, such as speed, GPS location, widths, heights, and categories of ships [160], [161], [162], [163]. Substantially, these data are highly heterogeneous compared to SAR data, due to the significantly different mechanisms for obtaining data. Optical images and AIS can provide abundant target information, and how to transfer the knowledge from them to SAR target recognition tasks is becoming a key research question.

## A. From Optical to SAR

Optical images can be divided into two categories: natural optical images and remote sensing optical images. Typically, natural optical images possess a considerable data volume, thereby offering ample supervised information. The remote sensing optical images always contain the targets of the same category in SAR images, which has a stronger correlation with target recognition. Due to the fact that optical images are more in line with the human visual system, it is easier to obtain annotations [32] compared with SAR images, which contain relatively blurry texture and speckle noise. Therefore, how to transfer the rich knowledge provided by optical images to SAR target recognition task to avoid annotating the SAR images, has attracted the interest of many researchers [32], [34], [86], [87], [88], [89], [90], [93], [94], [95], [97], [100], [101], [104], [106], [108], [110], [112], [113].

1) Pretraining Method: Compared to SAR images, optical images have a larger data volume and more diversity [37]. Pretraining on these data and then specializing the model for other downstream tasks in different domains, leveraging its ability to extract generalized features, is an efficient and effective approach [34], [86], [87], [88], [89], [90], [93], [94], [95], [97], [100], [101], [104], [106], [108].

Among these methods, pretrained on the ImageNet dataset or other natural optical image dataset, is the most common [34], [86], [93], [97]. However, Ying et al. [89] conducted pretraining experiments on both the MSTAR and ImageNet datasets. Their results demonstrated that pretraining the model on SAR images containing the same target led to better performance. Nevertheless, the performance obtained through pretraining on optical images was also noteworthy. To further improve the recognition performance, many pretraining methods combined with other technology are proposed [87], [88], [94], [100]. Lu et al. [87] utilized several data augmentation methods in the fine-tuning stage to improve the volume of the SAR images. Considering the pretrained model is easily over-fitting due to the scarce SAR image data in fine-tuning, Zhong et al. [88] proposed the method combined with the model compression technology to conduct better fine-tuning and accelerate the model's inference speed. The different resolutions of images bring difficulties to the ship recognition in SAR images, Relekar et al. [94] proposed a module integrated into the pretrained model to extract scalevariant features to improve the robustness of features. Wang et al. [100] believed that merely duplicating the single channel of SAR images three times to utilize a pretrained model that had been trained on optical images with three channels was obviously not reasonable. Instead, the subaperture decomposition algorithm [164] was employed to generate pseudocolor SAR images, allowing for better utilization of the general knowledge from optical images encoded in the pretrained model.

Although these methods have gained good performance on SAR target recognition, the targets in natural optical and SAR images exhibit a significant difference, which may harm the performance of the pretrained model. Several methods were proposed to perform pretraining on the optical land cover images [95] and the remote sensing optical ship images [90], [101]. These methods demonstrated that pretraining the model on remote sensing optical images leads to better performance. Moreover, the pretrained model has learned to effectively extract target features from remote sensing optical images, thereby enhancing SAR target recognition capabilities.

Different from the general pretrained method, Tai et al. [104] calculated network-layer-level and feature-channel-level transfer weights by attention mechanism to utilize the pretrained model more reasonably, thus achieving better performance on few-shot SAR target recognition. After that, Tai et al. [106] improved this work [104] by adding a push-attention mechanism and adjusting the learning rate according to the quality of SAR samples to conduct a better optimization strategy in the fine-tuning stage.

The difference between the imaging mechanisms of optical and SAR sensors indeed posed a significant challenge that constrains the improvement of models. Gao et al. [108] addressed this issue by introducing the SAR ship recognition method, leveraging a cross-modality GAN framework. They integrated the GAN with the attention mechanism CBAM [165] to generate robust pseudo-SAR images from optical remote sensing ship images. This strategy expanded the SAR image dataset, thereby enhancing the training process of the model.

Although pretraining is an effective technology used for TAL in SAR target recognition, there are still some directions worth studying. 1) The discrepancy between pretraining data and target SAR data, including data probability and modal discrepancy, is not fully considered. 2) The electromagnetic scattering features of SAR images are beneficial for target recognition, but they are often ignored in the pretraining methods.

2) Distribution Alignment Method: Due to the difference between the optical images and SAR images, there is a significant probability distribution discrepancy between them. The distribution alignment is a kind of effective method to solve this, and it is also commonly used in the field of TAL from optical to SAR [32], [110], [112].

Considering the long-tailed distributions of SAR image data, Chowdhury et al. [110] proposed to improve the performance of the method combined with knowledge distillation and class balancing. The target recognition knowledge learned from the electronic optical images was transferred to SAR target recognition by minimizing the KLD. However, the commonly used discrepancy measures, KLD and JSD, suffered the gradients vanish problem. They are not suitable for deep learning methods, which are based on first-order gradient optimization [32]. Mohammad et al. [32] used the Wasserstein distance (WD) to measure the distribution discrepancy between optical remote sensing images and SAR images. To reduce the computation burden, they approximated the WD with sliced WD (SWD) [166], which is the sum of multiple WD of 1-D distributions. It can be formulated by

$$SWD(P_S, P_T) = \int_{\mathbb{S}^{d-1}} \mathcal{W}(\mathcal{R}_{P_S}(\cdot; \gamma), \mathcal{R}_{P_T}(\cdot; \gamma)) d\gamma \quad (10)$$

where  $\mathcal{W}$  denotes the WD,  $\mathbb{S}^{d-1}$  denotes the *d*-dimensional sphere,  $\mathcal{R}_{P_S}(\cdot; \gamma)$  denotes the 1-D slice of the distribution along the direction  $\gamma$ . Therefore, as the feature distributions of remote sensing optical images and SAR images get closer, the classifier trained on the label remote sensing optical images and partially labeled SAR images can get better generalization.

The subdomain (subclass) distribution alignment is the same important, and the irrelevant information between the two domains will cause negative transfer learning. Zhao et al. [112] proposed a deep subdomain adaptation SAR ship recognition method. It measured the distribution discrepancy between optical remote sensing ship images and SAR ship images using LMMD, and integrated the CBAM to heuristically focus the network on the "what" and "where" for knowledge transfer learning, which is formulated as

$$\mathcal{L} = \mathcal{L}_C(x_S, y_S) + \text{LMMD}(x_S, x_T).$$
(11)

Distribution-alignment methods can obtain high performance, but there are still some questions worth studying:

- 1) The predefined distribution metrics may not be suitable for TAL from optical image to SAR image.
- Not all the features can be used to transfer, and how to decouple transferable features is a question worth considering.
- The discrepancy between the two modalities, optical image and SAR image, is not considered in these methods.

*3)* Shared Feature Space Method: The difference between shared-feature-space methods for homogeneous data and heterogeneous data lies in the fact that the latter aims to create a feature space, where features from different modalities cannot be distinguished.

Optical and SAR are two different modalities. If we directly transfer knowledge from optical images to SAR images cannot achieve satisfactory target recognition performance. Song



Fig. 8. Framework of cross-modality transfer learning.

et al. [113] proposed a two-stage cross-modality transfer learning ship recognition method. It first generated intermediate modality data from optical ship images using the GAN, then built a shared feature space to blend the intermediate domain and target domain through the adversarial learning method and aligned the distribution with LMMD, which can be seen in Fig. 8.

This method first combined shared-feature-space learning with GAN-based generation and distribution alignment. It is a novel integrated framework, which holds enlightening significance for subsequent SAR target recognition work.

#### B. From AIS to SAR

AIS contains abundant information specific to ships. Compared to optical images, AIS data are easier to obtain and have a larger volume. The length and width of the ship are common and transferable information in the AIS data [33], [116], [117], [118], [119]. How to design methods to utilize the naive geometric information from AIS to the SAR target recognition is a new topic recently.

1) Pretraining Method: Due to the real-time of AIS and its mandatory installation on ships of a certain tonnage, there is a large amount of available AIS data. Therefore, the pretraining technology is also used in TAL from AIS to SAR.

Yan et al. [116] pretrained the multiple ensemble classifier on the NGFs, calculated by the length and width in AIS data [33]. They extracted the lengths and widths of ships in SAR images with the minimum bounding box algorithm to construct SAR NGFs. Then, they utilized the pretrained classifier directly on SAR NGFs to obtain the categories of ships in SAR images.

There are still discrepancies between the NGFs from AIS data and SAR images, and directly using the pretrained model for SAR ship recognition is not reasonable. Lang et al. [33] proposed to perform model parameter regularization to further optimize the model, which is formulated as

$$\mathcal{L} = \frac{1}{2} \| w_T - \Gamma w_S \| + \lambda \sum_{i=1}^{n_T} \left( y_S^i \left( w_T x_T^i + b_T \right) - 1 \right)$$
(12)

where  $w_T$ ,  $b_T$  denote the weights and bias of the target SVM,  $w_S$  denote the weights and bias of the source SVM. They first trained a source SVM on the AIS NGFs. Then they froze it and trained the target SVM on SAR NGFs with L2-Regularization between the parameters of target SVM and source SVM. The target SVM can perform better on SAR ship recognition than directly using the pretrained model.

Although pretraining methods can achieve good performance, there are still some directions to be considered, such as the accuracy of NGFs, which is crucial but challenging to achieve in SAR images due to their blurry texture.

2) Distribution Alignment Method: Since NGFs can be directly obtained from AIS and through image processing algorithms from SAR images, there also exists a distribution discrepancy between the two types of NGFs. Some researchers have conducted early attempts to apply distribution alignment technology to achieve TAL from AIS data to SAR images.

Xu et al. [117] first proposed a framework, ARTL, which can be formulated as

$$\mathcal{L} = \mathcal{L}_{C} (x_{S}, y_{S}) + \sigma \mathcal{R} (\mathcal{F}) + \lambda \mathcal{R}_{d} (x_{S}, x_{T}) + \gamma \mathcal{R}_{m} (x_{S}, x_{T}) + \mu \mathcal{R}_{S} (x_{S})$$
(13)

where  $\sigma$ ,  $\lambda$ ,  $\gamma$ ,  $\mu$  denote the weights of regularization terms. By minimizing  $\mathcal{R}_d(x_S, x_T)$  between marginal distributions and conditional distributions, manifold consistency alignment  $\mathcal{R}_m(x_S, x_T)$  and source discriminative information preservation  $\mathcal{R}_S(x_S)$ , ARTL performed well in the semisupervised learning tasks for SAR ship recognition. After that, Xu et al. [118] proposed a method that combined metric learning, distribution alignment, and manifold regularization. It extended the method into semisupervised learning and unsupervised learning scenes for TAL from AIS to SAR images. The former methods achieved TAL with certain restrictions on using homogeneous features, which may hinder further improvement [119]. Yang et al. [119] first proposed semisupervised heterogeneous domain adaptation using a dynamic joint correlation alignment network. The whole loss function of it can be formulated as

$$\mathcal{L} = \mathcal{L}_C (x_S, y_S) + \mathcal{L}_C (x_T, y_T) + \alpha \left( (1 - \mu) \mathcal{L}_{\text{CORAL}} (x_S, x_T) + \mu \mathcal{L}_{\text{CORAL}} \left( (x_S, y_S), (x_T, y_T) \right) \right).$$
(14)

It transformed the AIS NGFs and SAR image into the same feature space to eliminate heterogeneity. In addition, it not only performed marginal and conditional distribution alignment, but also conducted pseudolabel refinement.

Distribution-alignment methods have also obtained satisfactory performance for TAL from AIS data to SAR images, but some directions still need to be explored: 1) The other features of SAR images besides NGFs were not utilized. 2) These methods did not consider the modality discrepancy between AIS and images.

#### C. From AIS and Optical to SAR

All previous references addressed the TAL from a single heterogeneous source domain, such as optical images or AIS data. Optical images and AIS data are totally different, providing texture visual features and NGFs, respectively. They are complementary to some extent. Therefore, developing a multisource heterogeneous transfer learning framework is a very valuable issue to be solved.

Lang et al. [102] proposed a multisource heterogeneous transfer learning method for SAR ship recognition. It first supplemented the feature vector with the same dimension zero vector of other domains. Then the multisource domains and target domain

Dataset	Resolution(m)	Chips	Categories
SAR Dataset			
MSTAR [42]	0.3  imes 0.3	5172(SOC), 1151(EOC1), 1212(EOC2)	BMP-2, BRDM-2, BTR-60, BTR-70, T-62, T-72, 2S1, ZSU-234, ZIL-131, D7
OpenSARShip [44]	$3 \times 22, 20 \times 22$	11346	Cargo, Tanker, Passenger, Law Enforcement, Anti-pollution equipment, Port Tender, Tug, Search and Rescue vessel, Pilot Vessel, High speed craft, Pleasure Craft, Diving ops, Dredging or underwater ops, Towing, Fishing, Wing in ground, other
FUSAR [62]	$1.124 \times (1.7 \sim 1.754)$	5243	type Cargo, Dive vessel, Dredger, Fishing, High speed craft, Law enforce, Passenger, Port tender, Reserved, Tanker, Wing in ground, SAR, Tug, Unspecified, Other
HR-SAR [103]	$2 \times 1.5$	450	Cargo, Bulk Carrier, Container, Tanker
MR-SAR [128]	$15 \times 15$	712	Bulk, Container, Oil Tanker
Simulation SAR Dat	taset		
SAMPLE [69]	0.3  imes 0.3	2690	2S1, BMP2, BTR70, M1, M2, M35, M548, M60, T72, ZSU234
Optical Dataset			
FGSCR [109]	-	9320	Aircraft Carrier, Cruiser, Destroyer, Assault Ship, Lading Ship, Transport Dock, Support Ship, Combat Ship, Frigate, Civil Vessel
HRSC2016 [114]	-	1061	Aircraft Carrier, Warcraft, Merchant Ship
AIS Dataset			
AIS-3 [33]	-	1500	Bulk Carriers, Containers, Tankers

 TABLE V

 Detailed Information of the Common Datasets

feature vectors were transformed into a common feature space to eliminate heterogeneity and excavate complementary across multisource domains, thus achieving knowledge transferring from multisource to SAR target recognition.

TAL from heterogeneous multisource domains to SAR image domain for target recognition is a completely new research field. Many directions need to be further considered:

- How to excavate the complementary and eliminate redundancy of multisource data.
- 2) How to overcome unbalanced amounts of multisource data.
- 3) How to eliminate the heterogeneity between multisource domains and between source domains and target domain.

## VI. AVAILABLE DATA

As analyzed in the previous sections, homogeneous and heterogeneous TALs aim to address distinct issues in specialized scenarios. For homogeneous TAL, the SAR datasets or the simulated SAR datasets are taken as the source domain. For heterogeneous transfer learning, the source domain may encompass the AIS dataset, optical image dataset, or multisource of AIS and optical image datasets. With the development of artificial intelligence in the remote sensing field, there are already many high-quality SAR image and optical image datasets available for target recognition. It is worth mentioning that there are relatively few high-quality available datasets for simulated SAR data and AIS. Many of the datasets used in previously reviewed literature are self-built datasets. In this section, we focus on the publicly available datasets commonly used in the reviewed literature, and introduce their configurations and usage scenarios. The detailed information on these common datasets can be seen in Table V. The unfilled cells in the second column (Resolution) of Table V indicate that the image resolution in this dataset is within a large range.

#### A. SAR Image Datasets

In this section, we focus on introducing the commonly used dataset, MSTAR [42], OpenSARShip [44], FUSAR [62], HR-SAR [103], MR-SAR [128], which can be used as source domain or target domain.

1) Vehicle Datasets: MSTAR [42] is the most commonly used vehicle target recognition dataset, which is collected by the Sandia National Laboratory and the U.S. Defense Advanced Research Projects Agency and the U.S. Air Force Research Laboratory. The images in MSTAR are obtained by a 10-GHz X-Band SAR satellite under different depression angles, with aspect views covering 0-360. The images have a resolution of  $0.3 \text{ m} \times 0.3 \text{ m}$  and a size of nearly 128 pixels  $\times$  128 pixels. It provides 190-300 images for the target in each depression angle. MSTAR can be extended into subdatasets under the standard operating condition (SOC) and various extent operating conditions (EOCs). SOC denotes the training images and test images are obtained at different depression angles with slight differences, such as 17° for training images and 15° for test images. EOCs denote that the training and test images are obtained at different depression angles with large differences, such as 17° for training images and 30° and 45° for test images, or at different noise levels, 1%, 5%, 10%, and 15% for test images.

2) Ship Datasets: The number of ship datasets is relatively larger than that of the vehicle dataset. The reviewed literature used many publicly available ship datasets [44], [62], [103], [128] and self-built ship datasets. OpenSARShip [44] is collected by Shanghai Jiao Tong University. The images in it are all obtained from a C-band Sentinel-1 satellite with polarization of VH and VV. It provides 11 346 ship chips from 41 SAR sea scene images, which contain 17 types of ships. FUSAR [62] is collected by Fudan University. The images in it are all obtained from China's first civil C-band Gaofen-3 satellite with polarizations of HH, HV, VH, and VV. It provides 5243 ship chips from

126 SAR images across a variety of scenarios, which include 15 primary ship categories and 98 subcategories. HR-SAR [103] is collected by the National University of Defense Technology, and the images in it are all obtained from TerraSAR-X with polarizations of HH, VH, and VV. It provides 450 ship chips from six scenes, encompassing three types of ships. MR-SAR [128] is collected by the Beijing University of Chemical Technology. The images in it are all obtained from eight scenes captured by Radarsat-2 with VV polarization, containing four types of ships and providing 712 chips for all types. In addition, some researchers collected the data from the private SAR images or extracted data from the ship detection datasets, HRSID [115], SSDD [167], SAR Ship dataset [99], LS-SSDD [168].

*3) Other SAR Datasets:* In addition to the target image described earlier, some public and self-built SAR scene classification datasets [169] are also used as the source domain for SAR characteristic knowledge transferring.

#### B. Simulated SAR Image Datasets

As the development of the computer simulated technology, some simulation tools are opened source, CASPatch [156], RaySAR [157], SARSIM [65]. Utilizing the electromagnetic imaging mechanism of the radar system along with the models of the target and its surrounding terrain, simulated SAR images could be generated through these tools. Therefore, the simulated SAR dataset is easier to obtain compared to the real SAR dataset. Nevertheless, there are relatively few publicly available datasets [69]. Many of the datasets used in these reviewed papers are self-built [170]. SAMPLE [69] is a commonly used simulated SAR vehicle dataset, which contains 1345 pairs of simulated and real SAR images. The real SAR vehicle images within it originate from the MSTAR. The simulation SAR vehicle images are derived by carefully crafting CAD models under configurations and sensor setting parameters aligned with the MSTAR. Notably, the vehicle categories represented in these simulation SAR images are identical to those in the MSTAR.

#### C. Optical Image Datasets

Due to the different imaging modes, optical images and SAR images are two completely heterogeneous data. Heterogeneous TAL methods leverage natural optical image datasets [171] to extract general discriminative features and employ remote sensing optical ship datasets [109], [172] to learn specific ship characteristics. The remote sensing optical ship images in these datasets are primarily sourced from Google Earth or various optical object detection datasets, including DOTA [173], HRSC2016 [114], and NWPUVHR-10 [174]. In addition, some datasets come from the competition data provided by IEEE [105] and Kaggle [91], [98].

## D. AIS Datasets

AIS is first designed for collision prevention, which exchanges static, dynamic, and voyage information between ships or between ship and the station. Given that AIS inherently includes information on the ship's category, downloading AIS data from public websites<sup>1</sup> is generally more convenient than manually annotating SAR images. Consequently, the majority of AIS data utilized in the reviewed literature originates from this.

Based on the available datasets mentioned, it is evident that there is a vast array of data that can be harnessed to verify the effectiveness of TAL methods for SAR target recognition. While some methods rely on self-built datasets, there are also standardized datasets that serve as valuable validation tools. For tank vehicle targets, the MSTAR and SAMPLE datasets are commonly utilized, while OpenSARShip, FUSAR, SAR Ship [103], and AIS-3 datasets focus on ship targets. Commonly employed configurations include: Validating the Homogeneous TAL method, such as MSTAR( $17^{\circ}$ )  $\rightarrow$  MSTAR( $15^{\circ}$ ), Open-SARShip  $\Leftrightarrow$  FUSAR, or SAMPLE  $\rightarrow$  MSTAR. Testing the Heterogeneous TAL method, for instance, FGSCR  $\rightarrow$  Open-SARShip or FUSAR, AIS-3 $\rightarrow$  OpenSARShip, FUSAR, or SAR Ship [103].

#### VII. EXPERIMENTS

In previous sections, we conducted a thorough analysis of diverse TAL algorithms tailored specifically for SAR target recognition tasks, along with the datasets utilized to validate the efficacy of these algorithms. In this section, we delve into the performance analysis of these TAL algorithms, and offer an intuitive evaluation and comparison of them with experiments. Specifically, we benchmarked them against general visual TAL methods designed for general target recognition, including DAN [175], DANN [176], DeepCoral [159], and DSAN [122], which are commonly employed in comparative experiments in the reviewed papers. The experiment was structured into two distinct groups to verify the algorithms for SAR land vehicle targets and SAR ocean ship targets, as these are currently the most common targets in SAR target recognition research.

#### A. Performance Evaluation Indexes

We evaluate the performance of SAR target recognition algorithms through average accuracy. Accuracy represents the percentage of correctly predicted samples among all samples, while average accuracy represents the average accuracy of all categories. Average accuracy can be formulated as

$$Precision_i = \frac{TP_i}{TP_i + FP_i}$$
(15a)

Average Accuracy = 
$$\frac{1}{c} \sum_{i=1}^{c} \operatorname{Precision}_{i}$$
 (15b)

where the Precision<sub>i</sub> represents the precision of class i, c represents the number of classes,  $TP_i$ ,  $FP_i$  represent the number of the samples correctly predicted as class i and the number of the samples incorrectly predicted as class i, and  $TP_i + FP_i$  represents the number of all samples as class i.

<sup>&</sup>lt;sup>1</sup>Marine Traffic website: http://www.marinetraffic.com/

References	Algorithms	Backbone	datasets	Accuracy
2017 Huang et al. [30]	pretraining	Self-built	SAR Scene⇒MSTAR SOC	99.05%
2018 Shang et al. [45]	pretraining	Self-built	Augmented MSTAR $\Rightarrow$ MSTAR SOC	99.71%
2023 Pei et al. [55]	Pretraining	ResNet50	Augmented MSTAR $\Rightarrow$ MSTAR SOC	99.88%
2022 Wang et al. [100]	Pretraining	A-ConvNet	ImageNet $\Rightarrow$ MSTAR SOC	99.13%
2021 Huang et al. [95]	Pretraining	Resnet18	ImageNet $\Rightarrow$ RESISC $\Rightarrow$ MSTAR SOC	99.46%
2022 Wang et al. [100]	Pretraining	ResNeXt50	ImageNet $\Rightarrow$ MSTAR SOC	99.88%
2022 Chen et al. [59]	Distribution Alignment	Resnet18	MSTAR SOC (20-shot)	99.96%
2021 Inkawhich et al. [68]	Pretraining	ResNet18	Augmented SAMPLE $\Rightarrow$ MSTAR	84.63%
2023 Kim et al. [72]	Pretraining	VGG16	$SAMPLE \Rightarrow GAN \Rightarrow MSTAR$	95.12%
2016 Sun et al. [159]	Distribution Alignment	Resnet50	$SAMPLE \Rightarrow MSTAR$	77.74%
2021 Zhu et al. [122]	Distribution Alignment	Resnet50	$SAMPLE \Rightarrow MSTAR$	87.95%
2023 Sun et al. [77]	Distribution Alignment	A-ConvNets	$SAMPLE \Rightarrow MSTAR$	94.94%
2023 Zhang et al. [79]	Distribution Alignment	proposed	$SAMPLE \Rightarrow MSTAR$	98.18%
2024 Han et al. [9]	Distribution Alignment	Resnet18	$SAMPLE \Rightarrow MSTAR$	96.08%
2015 Long et al. [175]	Shared Feature Space	Resnet50	$SAMPLE \Rightarrow MSTAR$	83.75%
2015 Ajakan et al. [176]	Shared Feature Space	Resnet50	$SAMPLE \Rightarrow MSTAR$	83.52%
2022 Chen et al. [12]	Shared Feature Space	Resnet18	$SAMPLE \Rightarrow MSTAR$	97.81%
2024 Shi et al. [84]	Shared Feature Space	Resnet50	$SAMPLE \Rightarrow MSTAR$	98.46%

 TABLE VI

 COMPARISON OF TAL METHODS FOR SAR VEHICLE TARGETS RECOGNITION

## B. Comparison of TAL Methods for SAR Vehicle Targets Recognition

The experimental results of TAL method for SAR land vehicle targets are presented in Table VI. In this table, MSTAR SOC denotes the data under SOCs of the MSTAR dataset, while MSTAR represents the MSTAR data corresponding to the SAM-PLE simulation dataset. Column 3 indicates backbone of the models used in these algorithms, with "self-built" indicating the backbone is specially designed. Column 5 displays the average accuracy of the algorithms.

From the upper part of Table VI, a comparative analysis of the experimental results under SOCs using MSTAR SOC data reveals that numerous pretraining techniques enhance model performance. This improvement is achieved either by acquiring pertinent knowledge from related domains or from augmented existing data. Notably, Huang et al. [30] enhanced the SAR target recognition models' performance by 0.75% compared with no pretraining (98.30%) by leveraging relevant knowledge derived from scene SAR images. Shang et al. [45] and Pei et al. [55] further optimized the model's performance through the utilization of augmented MSTAR data. Moreover, researchers have explored extracting relevant knowledge from general optical image datasets, which has significantly improved the model's recognition capabilities [95], [100]. It is worth noting that the transitive TAL of knowledge from general optical natural images to SAR scene images and then to SAR target images improved performance by 1.44% compared to directly transferring knowledge from general optical natural images to SAR target images (98.02%) [95]. Remarkably, Chen et al. [59] achieved nearperfect performance of 100% using methods tailored specifically for SAR vehicle targets, even under limited annotated sample conditions.

From the remaining part of Table VI, it becomes evident that numerous researchers aim to harness knowledge from simulated data to enhance the ability of their models in distinguishing real targets. Kim et al. [72] employed GANs to generate intermediate domain data from SAMPLE simulation data for pretraining, resulting in a remarkable 10.49% improvement over directly utilizing augmented SAMPLE simulation data [68]. Furthermore, the tailored TAL algorithms [77], [79] and [9] developed for SAR vehicle targets achieved superior performance, outperforming the general DSAN [122] by 6.99%, 10.23%, and 8.13%, respectively. Similarly, the shared feature space TAL algorithms [12] and [84] designed specifically for SAR targets demonstrated a significant boost in performance, surpassing the general adversarial TAL algorithm (DAN) by 14.06% and 14.71%, respectively. This underscores the effectiveness of these tailored algorithms in SAR vehicle target recognition tasks. In addition, algorithms [79] and [84], which utilized the electromagnetic scattering knowledge of SAR vehicles in their algorithm, achieved best performance in distribution alignment algorithms and shared feature space algorithms. It indicates that the models achieve better generalization by introducing electromagnetic scattering and traditional visual features.

After conducting numerous comparative experiments on SAR vehicle target recognition tasks, it has become evident that TAL algorithms specifically designed for SAR vehicle targets outperform general TAL methods. Furthermore, the generalization capability of the model can be significantly enhanced by incorporating electromagnetic scattering features or traditional visual features. However, it is worth noting that models trained using simulation data, like SAMPLE, exhibit lower generalization performance compared to those trained with real SAR datasets, such as MSTAR and SAR Scene datasets. This finding underscores the fact that the domain disparity between real SAR images is smaller than the disparity between simulated and real SAR images. Nevertheless, the accessibility of real data remains unparalleled compared to simulation data, making both scenarios worth of investigation.

## C. Comparison of TAL Methods for SAR Ship Targets Recognition

We validated the TAL algorithm for SAR ship recognition from optical images to SAR images and from AIS to SAR images using optical ship data, AIS data, and SAR ship data. The validation results of all algorithms can be seen from Table VII.

References	Algorithms	Backbone	datasets	Accuracy
2022 Lang et al. [101]	Pretraining	Self-built	Optical Ship $\Rightarrow$ SAR Ship	87.67%
2016 Sun et al. [159]	Distribution Alignment	Resnet50	Optical Ship $\Rightarrow$ SAR Ship	74.67%
2021 Zhu et al. [122]	Distribution Alignment	Resnet50	Optical Ship $\Rightarrow$ SAR Ship	84.00%
2022 Zhao et al. [112]	Distribution Alignment	Resnet50	Optical Ship $\Rightarrow$ SAR Ship	88.89%
2015 Ajakan et al. [176]	Shared Feature Space	Resnet50	Optical Ship $\Rightarrow$ SAR Ship	76.67%
2015 Long et al. [175]	Shared Feature Space	Resnet50	Optical Ship $\Rightarrow$ SAR Ship	72.00%
2018 Lang et al. [33]	Pretraining	SVM	$AIS \Rightarrow SAR Ship$	91.92%
2019 Xu et al. [117]	Distribution Alignment	SVM	$AIS \Rightarrow SAR Ship$	78.13%
2021 Xu et al. [118]	Distribution Alignment	1-NN	$AIS \Rightarrow SAR Ship$	75.33%
2022 Yang et al. [119]	Distribution Alignment	Self-built	AIS $\Rightarrow$ SAR Ship	83.18%
2022 Lang et al. [102]	Shared Feature Space	SVM	AIS+Optical Ship $\Rightarrow$ SAR Ship	85.60%

 TABLE VII

 COMPARISON OF TAL METHODS FOR SAR SHIP TARGETS RECOGNITION

From the upper part of Table VII, we can observe that the algorithm [112], specifically designed for SAR ship targets, outperforms the pretraining-based algorithm [101] by 1.22%. Furthermore, the algorithm [112] surpasses the recognition capabilities of the general TAL algorithm DSAN [122], which relies on distribution alignment, by a margin of 4.89%. In addition, it significantly outperforms the recognition performance of another general TAL algorithm, DAN [176], which is based on a shared feature space, by a significant 12.22%.

The lower part of the table reveals that certain methods designed specifically for transferring pertinent knowledge from AIS data to SAR images are highly effective for SAR ship target recognition. Notably, algorithm [119] achieved a performance of 83.18% solely by relying on the NGFs of ships provided by AIS. Remarkably, algorithm [102] enhances recognition performance by 2.42% by combining these NGFs from AIS with visual features derived from optical ship images.

After conducting comprehensive comparative experiments on SAR ship target recognition tasks, it becomes evident that the TAL algorithm tailored specifically for SAR ship targets outperforms general TAL methods. It is crucial to emphasize that the knowledge pertaining to ship targets, derived from both optical images and AIS data, can significantly boost the model's generalization capabilities. Furthermore, when these two data types are combined, they jointly provide pertinent knowledge, thereby further enhancing the model's generalization. This underscores the potential of multisource data to offer complementary information, ultimately elevating the performance of the model.

## VIII. DISCUSSION AND RECOMMENDATIONS

Existing TAL methods have succeeded in promising performance in SAR target recognition. There is still space for performance improvement, due to the gap between the performance of these methods and the performance upper bound. In addition, in most reviewed literature, it is supposed that there is unlabeled data (unsupervised) or partial data with annotations (semisupervised) in the target domain. However, some real scenarios in SAR target recognition do not match these settings. In this section, we provide some possible future directions for TAL in SAR target recognition.

## A. Challenge for Current Methods

TAL methods, whether homogeneous or heterogeneous, can be categorized into three main methods: pretraining, distribution alignment, and shared-feature-space approaches. These methods have different applicable scenarios and characteristics. However, they remain confronted with several challenges, such as insufficient data, difficulty in distribution discrepancy metric selection, feature discrimination maintenance, radar target characteristics utilization, and modality gap.

1) How to Overcome Insufficient Data: Pretraining methods necessitate an ample supply of relevant data to enable the model to acquire the ability to extract generalized features. The volumes of optical image data and AIS are relatively adequate due to the ease of data acquisition. However, there are almost no SAR images sufficient for pretraining. Generally speaking, the size and diversity of most datasets suitable for TAL in SAR target recognition are relatively limited, making it challenging to train a pretraining model with good generalization. As a result, there is an urgent need for well-organized, large-scale, and diverse benchmark datasets in the context of pretraining methods for SAR target recognition. In addition, generating more data utilizing generative artificial intelligence such as GAN [177] and diffusion models [178], or computer simulation techniques [65], [156], [157] are also possible to overcome the impact of insufficient data.

2) How to Select Distribution Discrepancy Metric: For distribution alignment methods, the most important issue is how to construct an appropriate metric of distribution discrepancy. First, the distribution alignment method excels in scenarios where the distribution differences are relatively small, such as TAL from simulated SAR to real SAR. However, in cross-modal TAL tasks, like from optical to SAR images, the predefined distribution alignment methods fall short in terms of performance. Furthermore, classification problems often encounter issues with unbalanced categories, whereas the distribution alignment method tends to overlook this aspect, disproportionately focusing on the transfer of categories with a larger sample size. Predefined metrics such as MMD, LMMD, KLD, SWD, and CORAL are commonly used in these methods. These metrics measure the distribution differences between the source and target domains from different perspectives. However, considering only marginal distribution, conditional distribution, or covariance alone is not comprehensive. Therefore, the challenge lies in selecting or designing comprehensive and suitable metrics for diverse scenarios within TAL for SAR target recognition, which remains a difficult and unexplored problem. Therefore, customized distribution discrepancy metrics tailored to specific scenarios, while taking into account imbalances in categories and long-tailed distributions, have the potential to overcome this challenge in the future.

3) How to Maintain Feature Discrimination: For shared-feature-space methods, most of them mapped the samples of source and target domains into domain-indivisible shared feature space with adversarial learning. However, it cannot guarantee the samples are mapped into property position in the shared feature space [179]. That is to say, it may reduce or damage the classification ability of the model, causing negative transfer. Therefore, it is worth studying how to map samples to the domain-indivisible shared feature spaces while maintaining or enhancing the discrimination of features. Integrating metric learning [180] and contrastive learning [181] to construct a shared feature space while preserving distinguishing features between classes can be a solution to address the scarcity of discrimination in the future.

4) How to Utilize Radar Target Characteristics and Traditional Features: Compared to other types of data, SAR images possess rich electromagnetic scatter characteristics that encompass the inherent properties of the radar targets themselves. However, there is insufficient utilization of SAR radar target electromagnetic characteristic information in knowledge transfer. In addition, traditional features exhibit strong stability and interpretability, making them nonnegligible. Therefore, considering the knowledge transfer of physical electromagnetic features and traditional features alongside visual features may further enhance the performance of recognition models. Alternatively, integrating these rich knowledge as prior information into TAL models may also be a good way to improve model generalization in the future.

5) How to Narrow the Modal Gap: In particular, the heterogeneous TAL is a problem involving cross-modality target recognition [37]. There is significant heterogeneity between SAR images and optical images or AIS, which was not considered in the method design. Overcoming these modal differences is a crucial step toward achieving effective heterogeneous TAL. How to separate modal-specific features and modal-invariant features through feature decoupling is worth considering, aiming to solve the heterogeneity in the process of knowledge transfer. Some advanced learning paradigms, methods, or models, such as contrastive learning [182], transformer [183], cross-modality translation [184], can be applied to overcome the heterogeneity in the future for TAL in SAR target recognition.

#### B. Future Directions

TAL is a hot topic with new technology coming out in an unending flow. Because of their relevance to some realistic scenarios of SAR target recognition, some meaningful points of TAL within the community deserve attention. Furthermore, the development of TAL in SAR target recognition could be accelerated by the application of advanced artificial intelligence



Fig. 9. Difference between closed set, partial, open set, and universal domain adaptation.

technologies, particularly the foundation model. Therefore, five points are summarized in this section to quickly understand the possible future directions of TAL in SAR target recognition.

1) Partial, Open-set, and Universal Domain Adaptation: The methods reviewed in this article primarily assume that the feature space of the source domain differs from that of the target domain, while the label space remains the same. This setup is usually referred to as the closed domain adaptation task  $(\mathcal{X}_S \neq \mathcal{X}_T, \mathcal{Y}_S = \mathcal{Y}_T)$ . However, this assumption poses significant limitations for realistic applications [185]. The methods have indeed addressed the issue of data shift between different domains, but they overlook the label shift (label spaces of the two domains not completely overlap) that also exists between these domains. This oversight is particularly problematic as the label shift is more in line with the realistic scenarios encountered in SAR target recognition. In real-world scenarios of TAL in SAR target recognition, various label space relationships can occur. These include cases where the target domain label space is a subset of the source domain label space ( $\mathcal{Y}_T \in \mathcal{Y}_S$ ), where the source domain label space is a subset of the target domain label space ( $\mathcal{Y}_S \in \mathcal{Y}_T$ ), where there is an intersection between the two label spaces ( $\mathcal{Y}_S \cap \mathcal{Y}_T \neq \emptyset$ ), or where the target domain label space is completely unknown ( $\mathcal{Y}_T = ?$ ). All these scenarios need to be taken into account when developing effective TAL methods for SAR target recognition, and the difference between these scenarios can be seen in Fig. 9.

Partial domain adaptation learning is designed to tackle the issue of transferring overlapping categories from the source domain to the target domain while avoiding the influence of redundant categories during knowledge transfer [186]. This approach ensures that only relevant information is transferred, thus enhancing the performance of target domain tasks. By focusing on the scenario of  $\mathcal{Y}_T \in \mathcal{Y}_S$ , partial domain adaptation learning aims to enhance the performance of target domain tasks by effectively filtering out irrelevant categories from the source domain.

*Open-set domain adaptation learning* is designed to address the issue of knowledge transfer from the source domain while distinguishing unknown categories in the target domain [187]. The key advantage of it is the ability to handle scenarios where the label spaces of the source and target domains are not fully aligned,  $\mathcal{Y}_S \in \mathcal{Y}_T$  or  $\mathcal{Y}_S \cap \mathcal{Y}_T \neq \emptyset$ , allowing for more generalizable models.

Universal domain adaptation learning is designed to solve the problem of correctly classifying known categories and distinguishing the unknown categories with the knowledge from the source domain [188]. Unlike open-set domain adaptation, which requires an intersection in the label space of the two domains, universal domain adaptation does not have this constraint. Consequently, the label space of the target domain remains unknown,  $\mathcal{Y}_T = ?$ .

The difficulty level of these four questions gradually rises, reflecting different scenarios encountered in SAR target recognition tasks. Here is a breakdown of each scenario:

- Closed domain adaptation: The target categories (target domain) to be identified are the same as the known categories (source domain).
- Partial domain adaptation: The known categories contain and exceed the categories of the targets to be identified.
- Open-set domain adaptation: The categories of the targets to be identified contain and exceed the known categories.
- 4) Universal domain adaptation: The relationship between the known categories and target categories is unknown.

These scenarios represent realistic challenges encountered in SAR target recognition tasks and are therefore worth of in-depth research to develop generalizable recognition systems.

2) Domain Generalization: When the operation conditions of radar, such as depression angle, polarization mode, wave band, etc., or the azimuth of the target suddenly change, or in some other situations, the data of the target domain are unavailable. That is to say, the distribution of data in the target domain is unseen. How to generalize the model to an invisible target domain under the condition of only source domain data (or multiple source domain data) available is a worthwhile research question, which may occur in real scenarios in the field of SAR target recognition. With the advancement of TAL technology, some researchers have proposed domain generalization techniques. These techniques aim at learning a model that can generalize to an unseen target domain, given one or more related source domain data [189], [190]. These techniques hold the potential to address the challenges mentioned above in the field of SAR target recognition [191], [192], [193].

3) Source-Free Domain Adaptation: Most of the reviewed methods need to revisit the data of the source domain during the knowledge transfer. However, due to the limitation of information security or communication bandwidth, the data of the source domain may not be allowed to be accessed [194], [195]. In this scenario, only the model trained on the data of the source domain is available, which is more in line with the information security requirements of SAR target recognition tasks. Source-free domain adaptation was introduced to overcome it relying only on a well-trained source model [196], [197]. This technology has the potential to be applied to the SAR target recognition tasks.

4) *Test-Time Adaptation:* When the platform undergoes continuous adjustments or the target undergoes continuous azimuth changes, the distribution of the data in the target domain also changes continuously. Every time the general method is built, we need to revisit the source domain data, which takes a lot of computational resources and slows down the learning efficiency of the method. In addition, the data of the source domain may not be visited due to information security [194], [195]. Some research indicated that generalizing the model to arbitrary distribution without the data of the target domain (domain generalization) is quite difficult [198], [199], [200]. Therefore, exploring how to maintain source domain knowledge while continuously learning from the target domain for model adjustment, without the need to revisit source domain data, is a worthwhile problem [60]. This is called test-time adaptation. Different from the sourcefree domain adaptation, test-time adaptation aims to perform adaptation in one small data batch during testing. Test-time adaptation alternates between training and testing during the testing process [201]. This technology is expected to solve the computational resource consumption problem caused by source domain data revisiting for TAL in SAR target recognition.

5) Foundation Models: In the past few years, there has been a flourishing era of foundation models. Foundation models typically use massive amounts of data for supervised or unsupervised learning, aiming to obtain general knowledge [41]. These neural networks always consist of billions or even hundreds of billions of parameters. With the increase of model parameters and data volume, the intelligence of foundation models has shown the emergence phenomenon [202]. The foundation model leverages the general knowledge acquired from massive data. By utilizing techniques such as transfer learning and incremental learning (learning new knowledge while preserving learned knowledge) [203], the model can be quickly adapted to various downstream tasks. These tasks are involved in domains like NLP and computer vision. In this way, a novel learning paradigm is established [204], [205], [206], [207], [208], [209].

Inspired by the development of the language foundation models and vision foundation models, some researchers adapt the state-of-the-art foundation models to the downstream tasks for remote sensing [210], [211], [212], [213], [214]. There are also some researchers dedicated to building and utilizing specialized remote sensing foundation models [215], [216], [217], [218], [219], [220].

What has milestone significance is the emergence of RingMo remote sensing foundation model [218]. Remote sensing data, which covers a wide range of tens of thousands of square kilometers and contains complex scene content, differs significantly from natural images used for building general foundation models. The RingMo remote sensing foundation model can be easily customized for downstream tasks [219], [220], indicating that the foundation model can also demonstrate extraordinary capabilities in the remote sensing field.

In the reviewed literature, the source domain is already given, which limits certain scenarios of SAR target recognition. In addition, the correlation between the source domain and the target domain greatly influences the difficulty of knowledge transfer. Nevertheless, the foundation model acquires knowledge from a wide and diverse range of data, and can effectively transfer general and robust target knowledge to downstream tasks based on the transfer learning paradigm. This promises to enable excellent performance in SAR target recognition.

#### IX. CONCLUSION

This survey offers a comprehensive overview of the latest advancements in TAL for target recognition in SAR images. Its primary aim is to consolidate knowledge and provide researchers with tools to quickly grasp the development of this field. We categorize the reviewed methods into homogeneous TAL and heterogeneous TAL, based on the relationship between the data types in the source and target domains. We further refine this categorization based on the specific data types used in the source domain and the underlying technologies. The reviewed studies suggested that TAL techniques hold tremendous potential in SAR target recognition tasks. Moreover, we examined the transferable traditional features of SAR images, which are notable for accurate target recognition based on TAL. In addition, we also reviewed the available datasets for validating the TAL method, and comparative experiments were conducted on these datasets to assess its performance. Finally, we proposed suggestions for enhancing current methods and discussed potential future research directions. Although TAL has demonstrated promising results in the field of SAR target recognition, there are still notable limitations to its application. 1) TAL relies on partially annotated data for pretraining or unlabeled data for distribution alignment and establishing a shared feature space. However, in SAR target recognition, the sample size of SAR data is often limited, hindering the full effectiveness of TAL. Furthermore, the presence of noise interference and incomplete target contours in SAR images can impact feature extraction, leading to suboptimal TAL performance. 2) TAL typically employs deep models and intricate algorithms, necessitating substantial computing resources and time. The long training and inference times associated with TAL methods may constrain their application in real-time SAR target recognition tasks, where speed and accuracy are crucial. 3) TAL is often built upon deep neural networks or other machine learning models, whose decision-making processes are often complex. In SAR target recognition, the reliability of model decisions is paramount, as incorrect identifications can have significant consequences. Therefore, ensuring the trustworthiness of TAL models is an area that requires further attention. We anticipate that future research will address these limitations of TAL methods for SAR target recognition, enabling more robust and efficient SAR target recognition systems. We also believe that the field of SAR target recognition leveraging TAL will continue to be an active and exciting area of research.

#### REFERENCES

- X. Yang, Q. Zhang, Q. Dong, Z. Han, X. Luo, and D. Wei, "Ship instance segmentation based on rotated bounding boxes for SAR images," *Remote Sens.*, vol. 15, Feb. 2023, Art. no. 1324.
- [2] X. Hu, Y. Wu, X. Liu, Z. Li, Z. Yang, and M. Li, "Intra- and inter-modal graph attention network and contrastive learning for SAR and optical image registration," *IEEE Trans. Geosci. Remote Sens.*, vol. 61, 2023, Art. no. 5220216.
- [3] Z. Yang, Y. Wu, M. Li, X. Hu, and Z. Li, "Unsupervised change detection in PolSAR images using Siamese encoder-decoder framework based on graph-context attention network," *Int. J. Appl. Earth Observ. Geoinf.*, vol. 124, Nov. 2023, Art. no. 103511.

- [4] M. Yang, X. Bai, L. Wang, and F. Zhou, "HENC: Hierarchical embedding network with center calibration for few-shot fine-grained SAR target classification," *IEEE Trans. Image Process.*, vol. 32, pp. 3324–3337, 2023.
- [5] Z. Liu, L. Wang, Z. Wen, K. Li, and Q. Pan, "Multilevel scattering center and deep feature fusion learning framework for SAR target recognition," *IEEE Trans. Geosci. Remote Sens.*, vol. 60, 2022, Art. no. 5227914.
- [6] K. Choi, T. Song, S. Kim, H. Jang, N. Ha, and K. Sohn, "Deep cascade network for noise-robust SAR ship detection with label augmentation," *IEEE Geosci. Remote Sens. Lett.*, vol. 19, 2022, Art. no. 4514005.
- [7] X. Yang, Q. Zhang, S. Zhao, D. Wei, and Z. Han, "Focal-pyramid-based vehicle segmentation in SAR images," *IEEE Geosci. Remote Sens. Lett.*, vol. 19, 2022, Art. no. 4028705.
- [8] L. Zhang, Y. Liu, W. Zhao, X. Wang, G. Li, and Y. He, "Frequencyadaptive learning for SAR ship detection in clutter scenes," *IEEE Trans. Geosci. Remote Sens.*, vol. 61, 2023, Art. no. 5215514.
- [9] F. Han, H. Dong, L. Si, and L. Zhang, "Improving SAR automatic target recognition via trusted knowledge distillation from simulated data," *IEEE Trans. Geosci. Remote Sens.*, vol. 62, 2024, Art. no. 5204314.
- [10] J. Liao, Y. Zhai, Q. Wang, B. Sun, and V. Piuri, "LDCL: Low-confidence discriminant contrastive learning for small-sample SAR ATR," *IEEE Trans. Geosci. Remote Sens.*, vol. 62, 2024, Art. no. 5202117.
- [11] Z. Wen, J. Liu, Z. Liu, S. Li, and Q. Pan, "Contrastive feature disentangling for partial aspect angles SAR noncooperative target recognition," *IEEE Trans. Geosci. Remote Sens.*, vol. 61, 2023, Art. no. 5103315.
- [12] Z. Chen, L. Zhao, Q. He, and G. Kuang, "Pixel-level and feature-level domain adaptation for heterogeneous SAR target recognition," *IEEE Geosci. Remote Sens. Lett.*, vol. 19, 2022, Art. no. 4515205.
- [13] Z. Wen, Z. Liu, S. Zhang, and Q. Pan, "Rotation awareness based self-supervised learning for SAR target recognition with limited training samples," *IEEE Trans. Image Process.*, vol. 30, pp. 7266–7279, 2021.
- [14] S. Zhao, Y. Xu, Y. Luo, W. Guo, B. Cai, and Z. Zhang, "A domain adaptation network for cross-imaging satellites SAR image ship classification," in *Proc. IEEE Int. Geosci. Remote Sens. Symp.*, Jul. 2022, pp. 1580–1583.
- [15] S. Zhao, Z. Zhang, T. Zhang, W. Guo, and Y. Luo, "Transferable SAR image classification crossing different satellites under open set condition," *IEEE Geosci. Remote Sens. Lett.*, vol. 19, 2022, Art. no. 4506005.
- [16] S. Zhao, Y. Luo, T. Zhang, W. Guo, and Z. Zhang, "Active learning SAR image classification method crossing different imaging platforms," *IEEE Geosci. Remote Sens. Lett.*, vol. 19, 2022, Art. no. 4514105.
- [17] L. Wang, X. Bai, R. Xue, and F. Zhou, "Few-shot SAR automatic target recognition based on Conv-BiLSTM prototypical network," *Neurocomputing*, vol. 443, pp. 235–246, Jul. 2021.
- [18] L. Wang, X. Yang, H. Tan, X. Bai, and F. Zhou, "Few-shot classincremental SAR target recognition based on hierarchical embedding and incremental evolutionary network," *IEEE Trans. Geosci. Remote Sens.*, vol. 61, 2023, Art. no. 5204111.
- [19] Z. Wang, L. Du, and Y. Li, "Boosting lightweight CNNs through network pruning and knowledge distillation for SAR target recognition," *IEEE J. Sel. Topics Appl. Earth Observ. Remote Sens.*, vol. 14, pp. 8386–8397, 2021.
- [20] M. Chendeb El Rai, J. H. Giraldo, M. Al-Saad, M. Darweech, and T. Bouwmans, "SemiSegSAR: A. semi-supervised segmentation algorithm for ship SAR images," *IEEE Geosci. Remote Sens. Lett.*, vol. 19, 2022, Art. no. 4510205.
- [21] J. Qu, X. Qiu, W. Wang, Z. Wang, B. Lei, and C. Ding, "A comparative study on classification features between high-resolution and polarimetric SAR images through unsupervised classification methods," *Remote Sens.*, vol. 14, Mar. 2022, Art. no. 1412.
- [22] L. Zhang and X. Gao, "Transfer adaptation learning: A decade survey," *IEEE Trans. Neural Netw. Learn. Syst.*, vol. 35, no. 1, pp. 23–44, Jan. 2024.
- [23] Q. Lang, L. Zhang, W. Shi, W. Chen, and S. Pu, "Exploring implicit domain-invariant features for domain adaptive object detection," *IEEE Trans. Circuits. Syst. Video Technol.*, vol. 33, no. 4, pp. 1816–1826, Apr. 2023.
- [24] B. Yang, J. Chen, X. Ma, and M. Ye, "Translation, association and augmentation: Learning cross-modality re-identification from single-modality annotation," *IEEE Trans. Image Process*, vol. 32, pp. 5099–5113, 2023.
- [25] D. Dong, G. Fu, J. Li, Y. Pei, and Y. Chen, "An unsupervised domain adaptation brain CT segmentation method across image modalities and diseases," *Expert. Sys. Appl.*, vol. 207, 2022, Art. no. 118016.

- [26] X. Zhang, Y. Li, Q. Pan, and C. Yu, "Triple loss adversarial domain adaptation network for cross-domain SEA-Land clutter classification," *IEEE Trans. Geosci. Remote Sens.*, vol. 61, 2023, Art. no. 5110718.
- [27] Y. Dong, T. Liang, C. Yang, H. Luo, and Y. Zhang, "Joint distance transfer metric learning for remote-sensing image classification," *IEEE Geosci. Remote Sens. Lett.*, vol. 19, 2022, Art. no. 6506205.
- [28] X. Zhang, Q. Pan, and S. García, "Multisource semisupervised adversarial domain generalization network for cross-scene Sea-Land clutter classification," 2024, arXiv:2402.06315.
- [29] L. Hu, H. Yu, W. Lu, D. Yin, X. Sun, and K. Fu, "AiRs: Adapter in remote sensing for parameter-efficient transfer learning," *IEEE Trans. Geosci. Remote Sens.*, vol. 62, 2024, Art. no. 5605218.
- [30] Z. Huang, Z. Pan, and B. Lei, "Transfer learning with deep convolutional neural network for SAR target classification with limited labeled data," *Remote Sens.*, vol. 9, Aug. 2017, Art. no. 907.
- [31] D. Malmgren-Hansen, A. Kusk, J. Dall, A. A. Nielsen, R. Engholm, and H. Skriver, "Improving SAR automatic target recognition models with transfer learning from simulated data," *IEEE Geosci. Remote Sens. Lett.*, vol. 14, no. 9, pp. 1484–1488, Sep. 2017.
- [32] M. Rostami, S. Kolouri, E. Eaton, and K. Kim, "Deep transfer learning for few-shot SAR image classification," *Remote Sens.*, vol. 11, Jun. 2019, Art. no. 1374.
- [33] H. Lang, S. Wu, and Y. Xu, "Ship classification in SAR images improved by AIS knowledge transfer," *IEEE Geosci. Remote Sens. Lett.*, vol. 15, no. 3, pp. 439–443, Mar. 2018.
- [34] M. A. Mufti, E. A. Hadhrami, B. Taha, and N. Werghi, "Automatic target recognition in SAR images: Comparison between pre-trained CNNs in a transfer learning based approach," in *Proc. Int. Conf. Artif. Intell. Big Data*, May 2018, pp. 160–164.
- [35] J. Li, Z. Yu, L. Yu, P. Cheng, J. Chen, and C. Chi, "A comprehensive survey on SAR ATR in deep-learning era," *Remote Sens.*, vol. 15, Mar. 2023, Art. no. 1454.
- [36] S. Zhao et al., "A review of single-source deep unsupervised visual domain adaptation," *IEEE Trans. Neural Netw. Learn. Syst.*, vol. 33, no. 2, pp. 473–493, Feb. 2022.
- [37] R. Bao et al., "A survey of heterogeneous transfer learning," 2023, arXiv:2310.08459.
- [38] K. El-Darymli, E. W. Gill, P. Mcguire, D. Power, and C. Moloney, "Automatic target recognition in synthetic aperture radar imagery: A state-of-the-art review," *IEEE Access*, vol. 4, pp. 6014–6058, 2016.
- [39] W. Jiang, Y. Wang, Y. Li, Y. Lin, and W. Shen, "Radar target characterization and deep learning in radar automatic target recognition: A review," *Remote Sens.*, vol. 15, Jul. 2023, Art. no. 3742.
- [40] G. Dong, G. Liao, H. Liu, and G. Kuang, "A review of the autoencoder and its variants: A comparative perspective from target recognition in synthetic-aperture radar images," *IEEE Geosci. Remote Sens. Mag.*, vol. 6, no. 3, pp. 44–68, Sep. 2018.
- [41] Y. Ma, S. Chen, S. Ermon, and D. B. Lobell, "Transfer learning in environmental remote sensing," *Remote Sens. Environ.*, vol. 301, Feb. 2024, Art. no. 113924.
- [42] E. R. Keydel, S. W. Lee, and J. T. Moore, "MSTAR extended operating conditions: A tutorial," in *Proc. SPIE Algorithms Synthetic Aperture Radar Imagery III*, Jun. 1996, vol. 2757, pp. 228–242.
- [43] D. Zhang, J. Liu, W. Heng, K. Ren, and J. Song, "Transfer learning with convolutional neural networks for SAR ship recognition," *Proc. IOP Conf. Ser. Mater. Sci. Eng.*, vol. 322, Mar. 2018, Art. no. 072001.
- [44] L. Huang et al., "OpenSARShip: A dataset dedicated to Sentinel-1 ship interpretation," *IEEE J. Sel. Topics Appl. Earth Observ. Remote Sens.*, vol. 11, no. 1, pp. 195–208, Jan. 2018.
- [45] R. Shang, J. Wang, L. Jiao, R. Stolkin, B. Hou, and Y. Li, "SAR targets classification based on deep memory convolution neural networks and transfer parameters," *IEEE J. Sel. Topics Appl. Earth Observ. Remote Sens.*, vol. 11, no. 8, pp. 2834–2846, Aug. 2018.
- [46] W. Zhang, Y. Zhu, and Q. Fu, "Semi-supervised deep transfer learningbased on adversarial feature learning for label limited SAR target recognition," *IEEE Access*, vol. 7, pp. 152412–152420, 2019.
- [47] Y. Zhai et al., "Robust SAR automatic target recognition based on transferred MS-CNN with L<sup>2</sup>-Regularization," *Comput. Intell. Neurosci.*, vol. 2019, Nov. 2019, Art. no. 9140167.
- [48] Y. Zhai et al., "MFFA-SARNET: Deep transferred multi-level feature fusion attention network with dual optimized loss for small-sample SAR ATR," *Remote Sens.*, vol. 12, Apr. 2020, Art. no. 1385.
- [49] C. Zhang et al., "Evaluation and improvement of generalization performance of SAR ship recognition algorithms," *IEEE J. Sel. Topics Appl. Earth Observ. Remote Sens.*, vol. 15, pp. 9311–9326, 2022.

- [50] M. Ma, J. Chen, W. Liu, and W. Yang, "Ship classification and detection based on CNN using GF-3 SAR images," *Remote Sens.*, vol. 10, Dec. 2018, Art. no. 2043.
- [51] C. Wang, H. Gu, and W. Su, "SAR image classification using contrastive learning and pseudo-labels with limited data," *IEEE Geosci. Remote Sens. Lett.*, vol. 19, 2022, Art. no. 4012505.
- [52] W. Zhu, H. Gu, and X. Zhu, "Synthetic aperture radar image classification based on constrictive learning with limited data," *IET Radar Sonar Navigation*, vol. 16, pp. 1530–1537, Sep. 2022.
- [53] Y. Liu, T. Li, K. Yuan, and S. Li, "Classifying ships in SAR images by using contour bias features and transfer learning," *IEEE Access*, vol. 10, pp. 129949–129957, 2022.
- [54] J. Liu, M. Xing, H. Yu, and G. Sun, "EFTL: Complex convolutional networks with electromagnetic feature transfer learning for SAR target recognition," *IEEE Trans. Geosci. Remote Sens.*, vol. 60, 2022, Art. no. 5209811.
- [55] H. Pei, M. Su, G. Xu, M. Xing, and W. Hong, "Self-supervised feature representation for SAR image target classification using contrastive learning," *IEEE J. Sel. Topics Appl. Earth Observ. Remote Sens.*, vol. 16, pp. 9246–9258, 2023.
- [56] Z. Huang, Z. Pan, and B. Lei, "What, where, and how to transfer in SAR target recognition based on deep CNNs," *IEEE Trans. Geosci. Remote Sens.*, vol. 58, no. 4, pp. 2324–2336, Apr. 2020.
- [57] L. Zhao, Q. He, D. Ding, S. Zhang, G. Kuang, and L. Liu, "Selecting pseudo supervision for unsupervised domain adaptive SAR target classification," *EURASIP J. Adv. Signal Process.*, vol. 2022, Sep. 2022, Art. no. 84.
- [58] Y. Zhang, X. Guo, H. Leung, and L. Li, "Transfer learning with shared and specific structures for SAR target recognition," in *Proc. IEEE Int. Geosci. Remote Sens. Symp.*, Jul. 2022, pp. 1003–1006.
- [59] K. Chen, Z. Pan, Z. Huang, Y. Hu, and C. Ding, "Learning from reliable unlabeled samples for semi-supervised SAR ATR," *IEEE Geosci. Remote Sens. Lett.*, vol. 19, 2022, Art. no. 4512205.
- [60] Y. Zhang, X. Guo, L. Li, and N. Ansari, "Deep knowledge integration of heterogeneous features for domain adaptive SAR target recognition," *Pattern Recognit.*, vol. 126, Jun. 2022, Art. no. 108590.
- [61] Y. Sun, Y. Wang, H. Liu, N. Wang, and J. Wang, "SAR target recognition with limited training data based on angular rotation generative network," *IEEE Geosci. Remote Sens. Lett.*, vol. 17, no. 11, pp. 1928–1932, Nov. 2020.
- [62] X. Hou, W. Ao, Q. Song, J. Lai, H. Wang, and F. Xu, "FUSAR-Ship: Building a high-resolution SAR-AIS matchup dataset of Gaofen-3 for ship detection and recognition," *Sci. China Inf. Sci.*, vol. 63, Mar. 2020, Art. no. 140303.
- [63] Y. Zhang, X. Guo, H. Leung, and L. Li, "Cross-task and cross-domain SAR target recognition: A meta-transfer learning approach," *Pattern Recognit.*, vol. 138, Jun. 2023, Art. no. 109402.
- [64] T. Gao, L. Gu, H. Chen, and B. Zhou, "Domain adaptation support tensor machine: An extended STM for object recognition using cross-source heterogeneous remote sensing data," *IEEE Trans. Geosci. Remote Sens.*, vol. 61, 2023, Art. no. 4507521.
- [65] J. M. Allan and M. J. Collins, "SARSIM: A digital SAR signal simulation system," in Proc. Remote Sens. Photogrammetry Soc., 2007, pp. 11–14.
- [66] L. Liu, Z. Pan, X. Qiu, and L. Peng, "SAR target classification with CycleGAN transferred simulated samples," in *Proc. IEEE Int. Geosci. Remote Sens. Symp.*, Jul. 2018, pp. 4411–4414.
- [67] Y. Ma, Y. Liang, W. Zhang, and S. Yan, "SAR target recognition based on transfer learning and data augmentation with LSGANs," in *Proc. Chin. Automat. Congr.*, Nov. 2019, pp. 2334–2337.
- [68] N. Inkawhich et al., "Bridging a GAP in SAR-ATR: Training on fully synthetic and testing on measured data," *IEEE J. Sel. Topics Appl. Earth Observ. Remote Sens.*, vol. 14, pp. 2942–2955, 2021.
- [69] B. P. Lewis, T. Scarnati, E. Sudkamp, J. W. Nehrbass, S. Rosencrantz, and E. G. Zelnio, "A SAR dataset for ATR development: The synthetic and measured paired labeled experiment (SAMPLE)," in *Proc. SPIE Algorithms Synth. Aperture Radar Imagery XXVI*, Apr. 2019, pp. 39–54.
- [70] W. Liang et al., "SAR target classification based on sample spectral regularization," *Remote Sens.*, vol. 12, Nov. 2020, Art. no. 3628.
- [71] L. Zhang et al., "Azimuth-aware discriminative representation learning for semi-supervised few-shot SAR vehicle recognition," *Remote Sens.*, vol. 15, Jan. 2023, Art. no. 331.
- [72] M. Kim and S. Kim, "Adaptation of synthetic SAR to measured SAR using complex value conditional GAN," in *Proc. Int. Conf. Control, Autom. Syst.*, Oct. 2023, pp. 241–244.

- [73] Y. Sun, Y. Wang, L. Hu, and H. Liu, "SAR target recognition using simulated data by an ensemble multi-scale deep domain adaptation recognition framework," in *Proc. IEEE Radar. Conf.*, Dec. 2021, pp. 1413–1417.
- [74] T. Tai and M. Toda, "Adapting intra-class variations for SAR image classification," in *Proc. Int. Conf. Image Process.*, Sep. 2021, pp. 2653–2657.
- [75] T. Tai and M. Toda, "Domain adaptation for SAR target recognition with limited training data via rigid transformation-based feature conversion," in *Proc. Digit. Int. Geosci. Remote Sens. Symp.*, Jul. 2022, pp. 247–250.
- [76] Q. He, L. Zhao, K. Ji, and G. Kuang, "SAR target recognition based on task-driven domain adaptation using simulated data," *IEEE Geosci. Remote Sens. Lett.*, vol. 19, 2022, Art. no. 4019205.
- [77] Y. Sun, Y. Wang, H. Liu, L. Hu, C. Zhang, and S. Wang, "Gradual domain adaptation with pseudo-label denoising for SAR target recognition when using only synthetic data for training," *Remote Sens.*, vol. 15, Jan. 2023, Art. no. 708.
- [78] X. Lv, X. Qiu, W. Yu, and F. Xu, "Simulation-aided SAR target classification via dual-branch reconstruction and subdomain alignment," *IEEE Trans. Geosci. Remote Sens.*, vol. 61, 2023, Art. no. 5214414.
- [79] C. Zhang, Y. Wang, H. Liu, Y. Sun, and S. Wang, "VSFA: Visual and scattering topological feature fusion and alignment network for unsupervised domain adaptation in SAR target recognition," *IEEE Trans. Geosci. Remote Sens.*, vol. 61, 2023, Art. no. 5216920.
- [80] W. Zhang, Y. Zhu, and Q. Fu, "Adversarial deep domain adaptation for multi-band SAR images classification," *IEEE Access*, vol. 7, pp. 78571–78583, 2019.
- [81] K. Wang, G. Zhang, and H. Leung, "SAR target recognition based on cross-domain and cross-task transfer learning," *IEEE Access*, vol. 7, pp. 153391–153399, 2019.
- [82] X. Lv, X. Qiu, and W. Yu, "Electromagnetic simulation aided SAR target classification via deep domain adaptation," in *Proc. IEEE Radar. Conf.*, Dec. 2021, pp. 995–998.
- [83] S. Du, J. Hong, Y. Wang, K. Xing, and T. Qiu, "Physical-related feature extraction from simulated SAR image based on the adversarial encoding network for data augmentation," *IEEE Geosci. Remote Sens. Lett.*, vol. 19, 2021, Art. no. 4016705.
- [84] Y. Shi, L. Du, C. Li, Y. Guo, and Y. Du, "Unsupervised domain adaptation for SAR target classification based on domain- and class-level alignment: From simulated to real data," *ISPRS J. Photogramm. Remote Sens.*, vol. 207, pp. 1–13, Jan. 2024.
- [85] "Imagenet," 2009. [Online]. Available: http://www.image-net.org/
- [86] Y. Wang, C. Wang, and H. Zhang, "Ship classification in high-resolution SAR images using deep learning of small datasets," *Sensors*, vol. 18, Sep. 2018, Art. no. 2929.
- [87] C. Lu and W. Li, "Ship classification in high-resolution SAR images via transfer learning with small training dataset," *Sensors*, vol. 19, Dec. 2019, Art. no. 63.
- [88] C. Zhong, X. Mu, X. He, J. Wang, and M. Zhu, "SAR target image classification based on transfer learning and model compression," *IEEE Geosci. Remote Sens. Lett.*, vol. 16, no. 3, pp. 412–416, Mar. 2019.
- [89] Z. Ying et al., "TAI-SARNET: Deep transferred atrous-inception CNN for small samples SAR ATR," Sensors, vol. 20, Mar. 2020, Art. no. 1724.
- [90] A. M. N. Taufique, N. Nagananda, and A. Savakis, "Visualization of deep transfer learning in SAR imagery," in *Proc. IEEE Int. Geosci. Remote Sens. Symp.*, Sep. 2020, pp. 3497–3500.
- [91] "Ships in satellite imagery—Kaggle," 2018. [Online]. Available: https: //www.kaggle.com/rhammell/ships-in-satellite-imagery
- [92] C. Schwegmann, W. Kleynhans, B. Salmon, L. Mdakane, and R. Meyer, "Very deep learning for ship discrimination in synthetic aperture radar imagery," in *Proc. IEEE Int. Geosci. Remote Sens. Symp.*, Jul. 2016, pp. 104–107.
- [93] R. Praneetha, T. Dhipu, and R. Rajesh, "SAR image classification using transfer learning," in *Proc. Int. Conf. Range Technol.*, Aug. 2021, pp. 1–6.
- [94] H. Relekar and P. Shanmugam, "Transfer learning based ship classification in Sentinel-1 images incorporating scale variant features," *Adv. Space Res.*, vol. 68, pp. 4594–4615, Dec. 2021.
- [95] Z. Huang, C. O. Dumitru, Z. Pan, B. Lei, and M. Datcu, "Classification of large-scale high-resolution SAR images with deep transfer learning," *IEEE Geosci. Remote Sens. Lett.*, vol. 18, no. 1, pp. 107–111, Jan. 2021.
- [96] G. Cheng, J. Han, and X. Lu, "Remote sensing image scene classification: Benchmark and state of the art," *Proc. IEEE*, vol. 105, no. 10, pp. 1865–1883, Oct. 2017.
- [97] B. W. Tienin, C. Guolong, and R. M. Esidang, "Comparative ship classification in heterogeneous dataset with pre-trained models," in *Proc. IEEE Radar. Conf.*, Mar. 2022, pp. 1–6.

- [98] "Airbus ship detection challenge—Kaggle," 2018. [Online]. Available: https://www.kaggle.com/c/airbus-ship-detection/overview
- [99] Y. Wang, C. Wang, H. Zhang, Y. Dong, and S. Wei, "A SAR dataset of ship detection for deep learning under complex backgrounds," *Remote Sens.*, vol. 11, Mar. 2019, Art. no. 765.
- [100] Z. Wang, X. Fu, and K. Xia, "Target classification for single-channel SAR images based on transfer learning with subaperture decomposition," *IEEE Geosci. Remote Sens. Lett.*, vol. 19, 2022, Art. no. 4003205.
- [101] H. Lang, R. Wang, S. Zheng, S. Wu, and J. Li, "Ship classification in SAR imagery by shallow CNN pre-trained on task-specific dataset with feature refinement," *Remote Sens.*, vol. 14, Nov. 2022, Art. no. 5986.
- [102] H. Lang, G. Yang, C. Li, and J. Xu, "Multisource heterogeneous transfer learning via feature augmentation for ship classification in SAR imagery," *IEEE Trans. Geosci. Remote Sens.*, vol. 60, 2022, Art. no. 5228814.
- [103] X. Xing, K. Ji, H. Zou, W. Chen, and J. Sun, "Ship classification in TerraSAR-X images with feature space based sparse representation," *IEEE Geosci. Remote Sens. Lett.*, vol. 10, no. 6, pp. 1562–1566, Nov. 2013.
- [104] Y. Tai, Y. Tan, S. Xiong, Z. Sun, and J. Tian, "Few-shot transfer learning for SAR image classification without extra SAR samples," *IEEE J. Sel. Topics Appl. Earth Observ. Remote Sens.*, vol. 15, pp. 2240–2253, 2022.
- [105] C. P. Schwegmann, W. Kleynhans, B. P. Salmon, L. W. Mdakane, and R. G. V. Meyer, "A SAR ship dataset for detection, discrimination and analysis," *IEEE Dataport*, 2017, doi: 10.21227/H2RK82.
- [106] Y. Tai, Y. Tan, S. Xiong, and J. Tian, "Cross-domain few-shot learning between different imaging modals for fine-grained target recognition," *IEEE J. Sel. Topics Appl. Earth Observ. Remote Sens.*, vol. 15, pp. 9186–9197, 2022.
- [107] H. Zhu, X. Chen, W. Dai, K. Fu, Q. Ye, and J. Jiao, "Orientation robust object detection in aerial images using deep convolutional neural network," in *Proc. IEEE Int. Conf. Image Process.*, Sep. 2015, pp. 3735–3739.
- [108] G. Gao, Y. Dai, X. Zhang, D. Duan, and F. Guo, "ADCG: A. crossmodality domain transfer learning method for synthetic aperture radar in ship automatic target recognition," *IEEE Trans. Geosci. Remote Sens.*, vol. 61, 2023, Art. no. 5109114.
- [109] Y. Di, Z. Jiang, and H. Zhang, "A public dataset for fine-grained ship classification in optical remote sensing images," *Remote Sens.*, vol. 13, Feb. 2021, Art. no. 747.
- [110] C. S. Jahan, A. Savakis, and E. Blasch, "SAR image classification with knowledge distillation and class balancing for long-tailed distributions," in *Proc. IEEE Image, Video, Multidimens. Signal Process. Workshop*, Jun. 2022, pp. 1–5.
- [111] J. Liu et al., "NTIRE 2021 multi-modal aerial view object classification challenge," in Proc. IEEE/CVF Conf. Comput. Vis. Pattern Recognit. Workshops, Jun. 2021, pp. 588–595.
- [112] S. Zhao and H. Lang, "Improving deep subdomain adaptation by dualbranch network embedding attention module for SAR ship classification," *IEEE J. Sel. Topics Appl. Earth Observ. Remote Sens.*, vol. 15, pp. 8038–8048, 2022.
- [113] Y. Song, J. Li, P. Gao, L. Li, T. Tian, and J. Tian, "Two-stage crossmodality transfer learning method for military-civilian SAR ship recognition," *IEEE Geosci. Remote Sens. Lett.*, vol. 19, 2022, Art. no. 4506405.
- [114] Z. Liu, L. Yuan, L. Weng, and Y. Yang, "A high resolution optical satellite image dataset for ship recognition and some new baselines," in *Proc. Int. Conf. Pattern Recognit. Appl. Methods*, Feb. 2017, pp. 324–331.
- [115] S. Wei, X. Zeng, Q. Qu, M. Wang, H. Su, and J. Shi, "HRSID: A. high-resolution SAR images dataset for ship detection and instance segmentation," *IEEE Access*, vol. 8, pp. 120234–120254, 2020.
- [116] Z. Yan, X. Song, L. Yang, and Y. Wang, "Ship classification in synthetic aperture radar images based on multiple classifiers ensemble learning and automatic identification system data transfer learning," *Remote Sens.*, vol. 14, Oct. 2022, Art. no. 5288.
- [117] Y. Xu, H. Lang, L. Niu, and C. Ge, "Discriminative adaptation regularization framework-based transfer learning for ship classification in SAR images," *IEEE Geosci. Remote Sens. Lett.*, vol. 16, pp. 1786–1790, Apr. 2019.
- [118] Y. Xu and H. Lang, "Ship classification in SAR images with geometric transfer metric learning," *IEEE Trans. Geosci. Remote Sens.*, vol. 59, no. 8, pp. 6799–6813, Aug. 2021.
- [119] G. Yang and H. Lang, "Semisupervised heterogeneous domain adaptation via dynamic joint correlation alignment network for ship classification in SAR imagery," *IEEE Geosci. Remote Sens. Lett.*, vol. 19, 2022, Art. no. 4508105.
- [120] S. J. Pan and Q. Yang, "A survey on transfer learning," *IEEE Trans. Knowl. Data Eng.*, vol. 22, no. 10, pp. 1345–1359, Oct. 2010.

- [121] L. Yang, Y. Liu, and W. Yang, "Transfer learning methods and their application in radar signal processing," in *Proc. Int. Conf. Big Data Artif. Intell.*, Jul. 2022, pp. 153–162.
- [122] Y. Zhu et al., "Deep subdomain adaptation network for image classification," *IEEE Trans. Neural Netw. Learn. Syst.*, vol. 32, no. 4, pp. 1713–1722, May 2021.
- [123] M. Gerry, L. Potter, I. Gupta, and A. Van Der Merwe, "A parametric model for synthetic aperture radar measurements," *IEEE Trans. Antennas Propag.*, vol. 47, no. 7, pp. 1179–1188, Jul. 1999.
- [124] F. Dellinger, J. Delon, Y. Gousseau, J. Michel, and F. Tupin, "SAR-SIFT: A SIFT-Like algorithm for SAR images," *IEEE Trans. Geosci. Remote Sens.*, vol. 53, no. 1, pp. 453–466, Jan. 2015.
- [125] N. Dalal and B. Triggs, "Histograms of oriented gradients for human detection," in *Proc. IEEE Comput. Soc. Conf. Comput. Vis. Pattern Recognit.*, Jun. 2005, vol. 1, pp. 886–893.
- [126] H. Wang, S. Li, Y. Zhou, and S. Chen, "SAR automatic target recognition using a roto-translational invariant wavelet-scattering convolution network," *Remote Sens.*, vol. 10, Mar. 2018, Art. no. 501.
- [127] H. Zhang, N. M. Nasrabadi, Y. Zhang, and T. S. Huang, "Multi-view automatic target recognition using joint sparse representation," *IEEE Trans. Aerosp. Electron. Syst.*, vol. 48, no. 3, pp. 2481–2497, Jul. 2012.
- [128] H. Lang and S. Wu, "Ship classification in moderate-resolution SAR image by naive geometric features-combined multiple kernel learning," *IEEE Geosci. Remote Sens. Lett.*, vol. 14, no. 10, pp. 1765–1769, Oct. 2017.
- [129] T. Zhang et al., "HOG-ShipCLSNet: A novel deep learning network with HOG feature fusion for SAR ship classification," *IEEE Trans. Geosci. Remote Sens.*, vol. 60, 2022, Art. no. 5210322.
- [130] J. B. Keller, "Geometrical theory of diffraction," J. Opt. Soc. Amer., vol. 52, pp. 116–130, Feb. 1962.
- [131] S. Feng, K. Ji, F. Wang, L. Zhang, X. Ma, and G. Kuang, "Electromagnetic scattering feature (ESF) module embedded network based on ASC model for robust and interpretable SAR ATR," *IEEE Trans. Geosci. Remote Sens.*, vol. 60, 2022, Art. no. 5235415.
- [132] S. Feng, K. Ji, F. Wang, L. Zhang, X. Ma, and G. Kuang, "PAN: Part attention network integrating electromagnetic characteristics for interpretable SAR vehicle target recognition," *IEEE Trans. Geosci. Remote Sens.*, vol. 61, 2023, Art. no. 5204617.
- [133] J. Zhou, S. Feng, H. Sun, L. Zhang, and G. Kuang, "Attributed scattering center guided adversarial attack for DCNN SAR target recognition," *IEEE Geosci. Remote Sens. Lett.*, vol. 20, 2023, Art. no. 4001805.
- [134] M. Gu, Y. Wang, H. Liu, and P. Wang, "PolSAR ship detection based on a SIFT-like PolSAR keypoint detector," *Remote Sens.*, vol. 14, Jun. 2022, Art. no. 2900.
- [135] G. Dong, H. Liu, and J. Chanussot, "Keypoint-based local descriptors for target recognition in SAR images: A comparative analysis," *IEEE Geosci. Remote Sens. Mag.*, vol. 9, no. 1, pp. 139–166, Mar. 2021.
- [136] G. Dong and H. Liu, "Radar target recognition via keypoint-based local descriptor in the context of Euclidean space and Riemannian manifold," *IEEE Trans. Instrum. Meas.*, vol. 70, 2021, Art. no. 2513010.
- [137] S. Song, B. Xu, and J. Yang, "SAR target recognition via supervised discriminative dictionary learning and sparse representation of the SAR-HOG feature," *Remote Sens.*, vol. 8, Aug. 2016, Art. no. 683.
- [138] J. Li, W. Xu, P. Shi, Y. Zhang, and Q. Hu, "LNIFT: Locally normalized image for rotation invariant multimodal feature matching," *IEEE Trans. Geosci. Remote Sens.*, vol. 60, 2022, Art. no. 5621314.
- [139] Y. Ye, L. Bruzzone, J. Shan, F. Bovolo, and Q. Zhu, "Fast and robust matching for multimodal remote sensing image registration," *IEEE Trans. Geosci. Remote Sens.*, vol. 57, no. 11, pp. 9059–9070, Nov. 2019.
- [140] H. Xie, J. He, Z. Lu, and J. Hu, "Two-level feature-fusion ship recognition strategy combining HOG features with dual-polarized data in SAR images," *Remote Sens.*, vol. 15, Sep. 2023, Art. no. 4393.
- [141] H. Lin, S. Song, and J. Yang, "Ship classification based on MSHOG feature and task-driven dictionary learning with structured incoherent constraints in SAR images," *Remote Sens.*, vol. 10, Jan. 2018, Art. no. 190.
- [142] X. Zhang, Y. Wang, D. Li, Z. Tan, and S. Liu, "Fusion of multifeature low-rank representation for synthetic aperture radar target configuration recognition," *IEEE Geosci. Remote Sens. Lett.*, vol. 15, no. 9, pp. 1402–1406, Sep. 2018.
- [143] R. Qin, X. Fu, J. Chang, and P. Lang, "Multilevel Wavelet-SRNet for SAR target recognition," *IEEE Geosci. Remote Sens. Lett.*, vol. 19, 2022, Art. no. 4009005.
- [144] P. Lang, X. Fu, C. Feng, J. Dong, R. Qin, and M. Martorella, "LW-CMDANet: A novel attention network for SAR automatic target recognition," *IEEE J. Sel. Topics Appl. Earth Observ. Remote Sens.*, vol. 15, pp. 6615–6630, 2022.

- [145] J. Wright, A. Y. Yang, A. Ganesh, S. S. Sastry, and Y. Ma, "Robust face recognition via sparse representation," *IEEE Trans. Pattern Anal. Mach. Intell.*, vol. 31, no. 2, pp. 210–227, Feb. 2009.
- [146] M. Elad and M. Aharon, "Image denoising via sparse and redundant representations over learned dictionaries," *IEEE Trans. Image Process.*, vol. 15, no. 12, pp. 3736–3745, Dec. 2006.
- [147] G. Dong, G. Kuang, N. Wang, and W. Wang, "Classification via sparse representation of steerable wavelet frames on Grassmann manifold: Application to target recognition in SAR image," *IEEE Trans. Image Process.*, vol. 26, no. 6, pp. 2892–2904, Jun. 2017.
- [148] A. Karine, A. Toumi, A. Khenchaf, and M. El Hassouni, "Target recognition in radar images using weighted statistical dictionary-based sparse representation," *IEEE Geosci. Remote Sens. Lett.*, vol. 14, no. 12, pp. 2403–2407, Dec. 2017.
- [149] M. Yu, S. Quan, G. Kuang, and S. Ni, "SAR target recognition via joint sparse and dense representation of monogenic signal," *Remote Sens.*, vol. 11, Nov. 2019, Art. no. 2676.
- [150] Z. Zhou, Z. Cao, and Y. Pi, "Subdictionary-based joint sparse representation for SAR target recognition using multilevel reconstruction," *IEEE Trans. Geosci. Remote Sens.*, vol. 57, no. 9, pp. 6877–6887, Sep. 2019.
- [151] M. Liu, S. Chen, J. Wu, F. Lu, X. Wang, and M. Xing, "SAR target configuration recognition via two-stage sparse structure representation," *IEEE Trans. Geosci. Remote Sens.*, vol. 56, no. 4, pp. 2220–2232, Apr. 2018.
- [152] E. Salerno, "Using low-resolution SAR scattering features for ship classification," *IEEE Geosci. Remote Sens. Lett.*, vol. 19, 2022, Art. no. 4509504.
- [153] W. Zhang, Y. Zhu, and Q. Fu, "Deep transfer learning based on generative adversarial networks for SAR target recognition with label limitation," in *Proc. IEEE Int. Conf. Signal, Inf. Data Process.*, Dec. 2019, pp. 1–5.
- [154] L. Berrada, A. Zisserman, and M. P. Kumar, "Smooth loss functions for deep top-k classification," in *Proc. Int. Conf. Learn. Represent.*, Feb. 2018, pp. 1–25.
- [155] Y. Jin, X. Wang, M. Long, and J. Wang, "Minimum class confusion for versatile domain adaptation," in *Proc. 16th Eur. Conf. Comput. Vis.*, Aug. 2020, pp. 464–480.
- [156] R. Zhang, J. Hong, and F. Ming, "CASpatch: A SAR image simulation code to support ATR applications," in *Proc. Asian-Pacific Conf. Synth. Aperture Radar*, Oct. 2009, pp. 502–505.
- [157] S. Auer, R. Bamler, and P. Reinartz, "RaySAR 3D SAR simulator: Now open source," in *Proc. IEEE Int. Geosci. Remote. Sens. Symp.*, Jul. 2016, pp. 6730–6733.
- [158] X. Xu, X. Zhou, R. Venkatesan, G. Swaminathan, and O. Majumder, "d-SNE: Domain adaptation using stochastic neighborhood embedding," in *Proc. IEEE Comput. Soc. Conf. Comput. Vis. Pattern Recognit.*, Jun. 2019, pp. 2492–2501.
- [159] B. Sun and K. Saenko, "Deep CORAL: Correlation alignment for deep domain adaptation," in *Proc. Eur. Conf. Comput. Vis.*, Sep. 2016, pp. 443–450.
- [160] D. Wright, C. Janzen, R. Bochenek, J. Austin, and E. Page, "Marine observing applications using AIS: Automatic identification system," *Front. Mar. Sci.*, vol. 6, Aug. 2019, Art. no. 537.
- [161] C. V. Ribeiro, A. Paes, and D. de Oliveira, "AIS-based maritime anomaly traffic detection: A review," *Expert Syst. Appl.*, vol. 231, Nov. 2023, Art. no. 120561.
- [162] S. Thombre et al., "Sensors and AI techniques for situational awareness in autonomous ships: A. review," *IEEE Trans. Intell. Transp. Syst.*, vol. 23, no. 1, pp. 64–83, Jan. 2022.
- [163] Z. Liu, J. Xu, J. Li, A. Plaza, S. Zhang, and L. Wang, "Moving ship optimal association for maritime surveillance: Fusing AIS and Sentinel-2 data," *IEEE Trans. Geosci. Remote Sens.*, vol. 60, 2022, Art. no. 5635218.
- [164] X. Leng, K. Ji, S. Zhou, and X. Xing, "Ship detection based on complex signal kurtosis in single-channel SAR imagery," *IEEE Trans. Geosci. Remote Sens.*, vol. 57, no. 9, pp. 6447–6461, Sep. 2019.
- [165] S. Woo, J. Park, J.-Y. Lee, and I. S. Kweon, "CBAM: Convolutional block attention module," in *Proc. Eur. Conf. Comput. Vis.*, 2018, pp. 3–19.
- [166] S. Kolouri, P. E. Pope, C. E. Martin, and G. K. Rohde, "Sliced Wasserstein auto-encoders," in *Proc. Int. Conf. Learn. Represent.*, May 2019, pp. 1–19.
- [167] J. Li, C. Qu, and J. Shao, "Ship detection in SAR images based on an improved faster R-CNN," in *Proc. SAR Big Data Era: Model., Methods Appl.*, Nov. 2017, pp. 1–6.
- [168] T. Zhang et al., "LS-SSDD-v1.0: A deep learning dataset dedicated to small ship detection from large-scale Sentinel-1 SAR images," *Remote Sens.*, vol. 12, Sep. 2020, Art. no. 2997.

- [169] C. O. Dumitru, G. Schwarz, and M. Datcu, "Land cover semantic annotation derived from high-resolution SAR images," *IEEE J. Sel. Topics Appl. Earth Observ. Remote Sens.*, vol. 9, no. 6, pp. 2215–2232, Jun. 2016.
- [170] A. Kusk, A. Abulaitijiang, and J. Dall, "Synthetic SAR image generation using sensor, terrain and target models," in *Proc. Eur. Conf. Syn. Aperture Radar*, Jun. 2016, pp. 1–5.
- [171] J. Deng, W. Dong, R. Socher, L.-J. Li, K. Li, and L. Fei-Fei, "ImageNet: A large-scale hierarchical image database," in *Proc. IEEE Comput. Soc. Conf. Comput. Vis. Pattern Recognit.*, Jun. 2009, pp. 248–255.
- [172] X. Zhang, Y. Lv, L. Yao, W. Xiong, and C. Fu, "A new benchmark and an attribute-guided multilevel feature representation network for finegrained ship classification in optical remote sensing images," *IEEE J. Sel. Topics Appl. Earth Observ. Remote Sens.*, vol. 13, pp. 1271–1285, 2020.
- [173] G.-S. Xia et al., "DOTA: A large-scale dataset for object detection in aerial images," in *Proc. IEEE Comput. Soc. Conf. Comput. Vis. Pattern Recognit.*, Jun. 2018, pp. 3974–3983.
- [174] G. Cheng, P. Zhou, and J. Han, "Learning rotation-invariant convolutional neural networks for object detection in VHR optical remote sensing images," *IEEE Trans. Geosci. Remote Sens.*, vol. 54, no. 12, pp. 7405–7415, Dec. 2016.
- [175] M. Long, Y. Cao, J. Wang, and M. I. Jordan, "Learning transferable features with deep adaptation networks," in *Proc. Int. Conf. Mach. Learn.*, 2015, pp. 97–105.
- [176] Y. Ganin, et al., "Domain-adversarial neural networks," J. Mach. Learn. Res., vol. 17, no. 1, pp. 2096–2030, Jan. 2016.
- [177] L. Li, C. Wang, H. Zhang, and B. Zhang, "SAR image ship object generation and classification with improved residual conditional generative adversarial network," *IEEE Geosci. Remote Sens. Lett.*, vol. 19, 2022, Art. no. 4000105.
- [178] T. An, B. Xue, C. Huo, S. Xiang, and C. Pan, "Efficient remote sensing image super-resolution via lightweight diffusion models," *IEEE Geosci. Remote Sens. Lett.*, vol. 21, 2024, Art. no. 6000105.
- [179] M. H. Zonoozi and V. Seydi, "A survey on adversarial domain adaptation," *Neural Process Lett.*, vol. 55, pp. 2429–2469, Aug. 2022.
- [180] Y. Li, X. Li, Q. Sun, and Q. Dong, "SAR image classification using CNN embeddings and metric learning," *IEEE Geosci. Remote Sens. Lett.*, vol. 19, 2022, Art. no. 4002305.
- [181] P. H. Le-Khac, G. Healy, and A. F. Smeaton, "Contrastive representation learning: A framework and review," *IEEE Access*, vol. 8, pp. 193907–193934, 2020.
- [182] C. Li, W. Guo, Z. Zhang, and T. Zhang, "Self-supervised classification of SAR images with optical image assistance," *IEEE Trans. Geosci. Remote Sens.*, vol. 61, 2023, Art. no. 5221715.
- [183] A. Vaswani et al., "Attention is all you need," in Proc. 31st Int. Conf. Neural Inf. Process. Syst., 2017, pp. 6000–6010.
- [184] H. Li, C. Gu, D. Wu, G. Cheng, L. Guo, and H. Liu, "Multiscale generative adversarial network based on wavelet feature learning for SAR-to-Optical image translation," *IEEE Trans. Geosci. Remote Sens.*, vol. 60, 2022, Art. no. 5236115.
- [185] K. You, M. Long, Z. Cao, J. Wang, and M. I. Jordan, "Universal domain adaptation," in *Proc. IEEE Comput. Soc. Conf. Comput. Vis. Pattern Recognit.*, Jun. 2019, pp. 2715–2724.
- [186] J. Zhang, Z. Ding, W. Li, and P. Ogunbona, "Importance weighted adversarial nets for partial domain adaptation," in *Proc. IEEE Comput. Soc. Conf. Comput. Vis. Pattern Recognit.*, Jun. 2018, pp. 8156–8164.
- [187] X. Ma, K. Ji, L. Zhang, S. Feng, B. Xiong, and G. Kuang, "An open set recognition method for SAR targets based on multitask learning," *IEEE Geosci. Remote Sens. Lett.*, vol. 19, 2022, Art. no. 4014005.
- [188] K. You, M. Long, Z. Cao, J. Wang, and M. I. Jordan, "Universal domain adaptation," in *Proc. IEEE Comput. Soc. Conf. Comput. Vis. Pattern Recognit.*, Jan. 2019, pp. 2715–2724.
- [189] J. Wang et al., "Generalizing to unseen domains: A survey on domain generalization," *IEEE Trans. Knowl. Data Eng.*, vol. 35, no. 8, pp. 8052–8072, Aug. 2023.
- [190] K. Zhou, Z. Liu, Y. Qiao, T. Xiang, and C. C. Loy, "Domain generalization: A survey," *IEEE Trans. Pattern Anal. Mach. Intell.*, vol. 45, no. 4, pp. 4396–4415, Apr. 2023.
- [191] Y. Zhang, W. Li, W. Sun, R. Tao, and Q. Du, "Single-source domain expansion network for cross-scene hyperspectral image classification," *IEEE Trans. Image Process.*, vol. 32, pp. 1498–1512, 2023.

- [192] Z. Gao, B. Pan, X. Xu, T. Li, and Z. Shi, "LiCa: Label-indicateconditional-alignment domain generalization for pixel-wise hyperspectral imagery classification," *IEEE Trans. Geosci. Remote Sens.*, vol. 61, 2023, Art. no. 5519011.
- [193] H. Zhao, J. Zhang, L. Lin, J. Wang, S. Gao, and Z. Zhang, "Locally linear unbiased randomization network for cross-scene hyperspectral image classification," *IEEE Trans. Geosci. Remote Sens.*, vol. 61, 2023, Art. no. 5526512.
- [194] J. Drozdowicz, "The open-source framework for 3D synthetic aperture radar simulation," *IEEE Access*, vol. 9, pp. 39518–39529, 2021.
- [195] N. Ma, H. Wang, Z. Zhang, S. Zhou, H. Chen, and J. Bu, "Source-free semi-supervised domain adaptation via progressive mixup," *Knowl.-Based Syst.*, vol. 262, Feb. 2023, Art. no. 110208.
- [196] Y. Liu, C. Zhao, Y. Lu, W. Xing, and X. Qiao, "A source free domain adaptation model based on adversarial learning for image classification," *Appl. Intell.*, vol. 53, pp. 11389–11402, May 2023.
- [197] L. Zhou, N. Li, M. Ye, X. Zhu, and S. Tang, "Source-free domain adaptation with class prototype discovery," *Pattern Recognit.*, vol. 145, Feb. 2024, Art. no. 109974.
- [198] Y.-F. Zhang et al., "AdaNPC: Exploring non-parametric classifier for test-time adaptation," in *Proc. Mach. Learn. Res.* Apr. 2023, pp. 41647– 41676.
- [199] Y.-F. Zhang et al., "Domain-specific risk minimization for domain generalization," in *Proc. ACM SIGKDD Int. Conf. Knowl. Discov. Data Min.*, Aug. 2023, pp. 3409–3421.
- [200] A. Dubey, V. Ramanathan, A. Pentland, and D. Mahajan, "Adaptive methods for real-world domain generalization," in *Proc. IEEE Comput. Soc. Conf. Comput. Vis. Pattern Recognit.*, Jun. 2021, pp. 14335–14344.
- [201] Q. Wang, O. Fink, L. Van Gool, and D. Dai, "Continual test-time domain adaptation," in *Proc. IEEE Comput. Soc. Conf. Comput. Vis. Pattern Recognit.*, Jun. 2022, pp. 7201–7211.
- [202] J. Wei et al., "Emergent abilities of large language models," 2022, arXiv:2206.07682.
- [203] M. Masana, X. Liu, B. Twardowski, M. Menta, A. D. Bagdanov, and J. van de Weijer, "Class-incremental learning: Survey and performance evaluation on image classification," *IEEE Trans. Pattern Anal. Mach. Intell.*, vol. 45, no. 5, pp. 5513–5533, May 2023.
- [204] L. Yuan et al., "Florence: A new foundation model for computer vision," 2021, arXiv:2111.11432.
- [205] H. Bao, L. Dong, S. Piao, and F. Wei, "BEiT: BERT pre-training of image transformers," in *Proc. Int. Conf. Learn. Represent.*, Jan. 2022, pp. 1–18.
- [206] G. Mai et al., "On the opportunities and challenges of foundation models for geospatial artificial intelligence," ACM Trans. Spatial Algorithms Syst., vol. 10, no. 2, Art. no. 11, Jul. 2024.
- [207] L. Zhao et al., "When brain-inspired AI meets AGI," Meta-Radiol., vol. 1, Jun. 2023, Art. no. 100005.
- [208] Y. Liu et al., "Summary of ChatGPT-Related research and perspective towards the future of large language models," *Meta- Radiol.*, vol. 1, Sep. 2023, Art. no. 100017.
- [209] A. Kirillov et al., "Segment anything," in Proc. IEEE Int. Conf. Comput. Vision, Oct. 2023, pp. 3992–4003.
- [210] L. P. Osco et al., "The segment anything model (SAM) for remote sensing applications: From zero to one shot," *Int. J. Appl. Earth Observ. Geoinf.*, vol. 124, Nov. 2023, Art. no. 103540.
- [211] X. Liu, "A SAM-based method for large-scale crop field boundary delineation," in Proc. Annu. IEEE Commun. Soc. Conf. Sens., Mesh Ad Hoc Commun. Netw. Workshops, Sep. 2023, pp. 1–6.
- [212] F. Liu et al., "RemoteCLIP: A vision language foundation model for remote sensing," in *IEEE Trans. Geosci. Remote Sens.*, vol. 62, 2024, Art no. 5622216.
- [213] Y. Hu, J. Yuan, C. Wen, X. Lu, and X. Li, "RSGPT: A remote sensing vision language model and benchmark," 2023, arXiv:2307.15266.
- [214] X. He, Y. Chen, L. Huang, D. Hong, and Q. Du, "Foundation model-based multimodal remote sensing data classification," *IEEE Trans. Geosci. Remote Sens.*, vol. 62, 2024, Art. no. 5502117.
- [215] D. Wang et al., "Advancing plain vision transformer toward remote sensing foundation model," *IEEE Trans. Geosci. Remote Sens.*, vol. 61, 2023, Art. no. 5607315.
- [216] K. Cha, J. Seo, and T. Lee, "A billion-scale foundation model for remote sensing images," *IEEE J. Sel. Topics Appl. Earth Observ. Remote Sens.* 2024, doi: 10.1109/JSTARS.2024.3401772.

- [217] L. Jiao et al., "Brain-inspired remote sensing foundation models and open problems: A comprehensive survey," *IEEE J. Sel. Topics Appl. Earth Observ. Remote Sens.*, vol. 16, pp. 10084–10120, 2023.
- [218] X. Sun et al., "RingMo: A. remote sensing foundation model with masked image modeling," *IEEE Trans. Geosci. Remote Sens.*, vol. 61, 2023, Art. no. 5612822.
- [219] F. Yao et al., "RingMo-Sense: Remote sensing foundation model for spatiotemporal prediction via spatiotemporal evolution disentangling," *IEEE Trans. Geosci. Remote Sens.*, vol. 61, 2023, Art. no. 5620821.
- [220] Z. Yan et al., "RingMo-SAM: A foundation model for segment anything in multimodal remote-sensing images," *IEEE Trans. Geosci. Remote Sens.*, vol. 61, 2023, Art. no. 5625716.



Xinpeng Yang received the B.S. degree in measurement and control technology and instrument from the School of Information Science and Engineering, Shenyang University of Technology, Shenyang, China, in 2020, and the M.S. degree in newgeneration electronic information technology from the School of Electronic Engineering, Xidian University, Xi'an, China, in 2023. He is currently working toward the Ph.D. degree in control theory and control engineering with the School of Automation, Northwestern Polytechnical University, Xi'an.

His main research direction is radar automatic target detection and recognition, and transfer learning.



Lianmeng Jiao (Member, IEEE) received the B.E. and M.E. degrees in control science and engineering from Northwestern Polytechnical University, Xi'an, China, in 2009 and 2012, respectively, and the Ph.D. degree in computer science from the Université de Technologie de Compiègne, Compiègne, France, in 2015.

Since 2019, he has been an Associate Professor with the School of Automation, Northwestern Polytechnical University. He has authored three books and more than 50 peer-reviewed journals and conference

papers. His research interests include machine learning, data mining, information fusion, belief function theory and their application in pattern recognition and decision making.

Dr. Jiao has been the Session Chairs of ICIUS'2016/CCC'2021/BELIEF' 2021/ICBDT'2022/PRAI'2023, an Organization Committee Member of BFAS'2017, a Guest Editor of the Entropy and CMES Journals, and the Editorial Board Member of several peer reviewed journals including PLOS One and *American Journal of Artificial Intelligence*.



**Quan Pan** was born in China in 1961. He received the B.S. degree in automatic control from the Huazhong University of Science and Technology, Wuhan, China, in 1982, and the M.S. and Ph.D. degrees in control theory and applications from Northwestern Polytechnical University, Xi'an, China, in 1991 and 1997, respectively.

From 1991 to 1993, he was an Associate Professor with Northwestern Polytechnical University, where he has been a Professor with the Department of Automatic Control since 1997. He has authored 11 books

and more than 400 articles. His research interests include information fusion, target tracking and recognition, deep network and machine learning, unmanned aerial vehicle detection navigation and security control, polarization spectral imaging and image processing, industrial control system information security, commercial password applications, and modern security technologies.

Dr. Pan is an Associate Editor of the Journal of Information Fusion and Modern Weapons Testing Technology.